WEIL, GOTSHAL & MANGES LLP 767 Fifth Avenue New York, New York 10153 Telephone: (212) 310-8000 Facsimile: (212) 310-8007 Ralph I. Miller Peter Gruenberger Jacqueline Marcus

Attorneys for Debtors and Debtors in Possession

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF NEW YORK

	X	
	:	
In re	:	Chapter 11 Case No.
	:	
LEHMAN BROTHERS HOLDINGS INC., et al.,	:	08-13555 (JMP)
	:	
Debtors.	:	(Jointly Administered)
	:	
	:	
	X	

NOTICE OF DEBTORS' MOTION PURSUANT TO RULE 9019 OF THE FEDERAL RULES OF BANKRUPTCY PROCEDURE FOR APPROVAL OF A SETTLEMENT AMONG LEHMAN BROTHERS SPECIAL FINANCING INC., BNY CORPORATE TRUSTEE SERVICES LIMITED, PERPETUAL TRUSTEE COMPANY LIMITED, AND OTHERS, RELATING TO CERTAIN SWAP <u>TRANSACTIONS WITH SAPHIR FINANCE PUBLIC LIMITED COMPANY</u>

PLEASE TAKE NOTICE that a hearing on the annexed Motion of Lehman Brothers Special Financing Inc. and its affiliated debtors in the above-referenced chapter 11 cases, including Lehman Brothers Holdings Inc. (collectively, the "<u>Debtors</u>") pursuant to Rule 9019 of the Federal Rules of Bankruptcy Procedure, for approval of a settlement among LBSF, LBHI, BNY Corporate Trustee Services Limited, Perpetual Trustee Company Limited, and others, all as more fully described in the Motion, will be held before the Honorable James M. Peck, United States Bankruptcy Judge, at the United States Bankruptcy Court, Alexander Hamilton Customs House, Courtroom 601, One Bowling Green, New York, New York 10004 (the "<u>Bankruptcy Court</u>"), on **December 15, 2010 at 10:00 a.m. (Prevailing Eastern Time)** (the "<u>Hearing</u>").

PLEASE TAKE FURTHER NOTICE that objections, if any, to the Motion shall be in writing, shall conform to the Federal Rules of Bankruptcy Procedure (the "<u>Bankruptcy</u> <u>Rules</u>") and the Local Rules of the Bankruptcy Court for the Southern District of New York, shall set forth the name of the objecting party, the basis for the objection and the specific grounds

thereof, shall be filed with the Bankruptcy Court electronically in accordance with General Order M-242 (which can be found at www.nysb.uscourts.gov) by registered users of the Bankruptcy Court's case filing system and by all other parties in interest, on a 3.5 inch disk, preferably in Portable Document Format (PDF), WordPerfect, or any other Windows-based word processing format (with two hard copies delivered directly to Chambers), and shall be served upon: (i) the chambers of the Honorable James M. Peck, One Bowling Green, New York, New York 10004, Courtroom 601; (ii) Weil Gotshal & Manges LLP, 767 Fifth Avenue, New York, New York 10153, Attn: Ralph I. Miller, Esq., Peter Gruenberger, Esq., and Jacqueline Marcus, Esq., attorneys for the Debtors; (iii) the Office of the United States Trustee for the Southern District of New York (the "U.S. Trustee"), 33 Whitehall Street, 21st Floor, New York, New York 10004 Attn: Tracy Hope Davis, Esq., Andy Velez-Rivera, Esq., Paul Schwartzberg, Esq., Brian Masumoto, Esq., and Linda Riffkin, Esq.; and (iv) Milbank, Tweed, Hadley & McCloy LLP, 1 Chase Manhattan Plaza, New York, New York 10005, Attn: Dennis F. Dunne, Esq., Evan Fleck, Esq., and Dennis O'Donnell, Esq., attorneys for the Official Committee of Unsecured Creditors appointed in these cases; and (v) Reed Smith LLP, 225 Fifth Avenue, Pittsburgh, Pennsylvania 15222, Attn: Eric A. Schaffer, Esq., so as to be so filed and received by no later than December 8, 2010 at 4:00 p.m. (prevailing Eastern Time) (the "Objection Deadline").

PLEASE TAKE FURTHER NOTICE that if an objection to the Motion is not received by the Objection Deadline, the relief requested shall be deemed unopposed, and the Bankruptcy Court may enter an order granting the relief sought without a hearing.

PLEASE TAKE FURTHER NOTICE that objecting parties are required to attend the Hearing, and failure to appear may result in relief being granted or denied upon default.

Dated: November 24, 2010 New York, New York

> <u>/s/ Jacqueline Marcus</u> Ralph I. Miller Peter Gruenberger Jacqueline Marcus

WEIL, GOTSHAL & MANGES LLP 767 Fifth Avenue New York, New York 10153 Telephone: (212) 310-8000 Facsimile: (212) 310-8007

Attorneys for Debtors and Debtors in Possession

WEIL, GOTSHAL & MANGES LLP 767 Fifth Avenue New York, New York 10153 Telephone: (212) 310-8000 Facsimile: (212) 310-8007 Ralph I. Miller Peter Gruenberger Jacqueline Marcus

Attorneys for Debtors and Debtors in Possession

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF NEW YORK

-----X

	X	
	:	
In re	:	Chapter 11 Case No.
	:	
LEHMAN BROTHERS HOLDINGS INC., et al.,	:	08-13555 (JMP)
	:	
Debtors.	:	(Jointly Administered)
	:	
	:	

DEBTORS' MOTION PURSUANT TO RULE 9019 OF THE FEDERAL RULES OF BANKRUPTCY PROCEDURE FOR APPROVAL OF A SETTLEMENT AMONG LEHMAN BROTHERS SPECIAL FINANCING INC., BNY CORPORATE TRUSTEE SERVICES LIMITED, PERPETUAL TRUSTEE COMPANY LIMITED, AND OTHERS, RELATING TO CERTAIN SWAP TRANSACTIONS WITH SAPHIR FINANCE PUBLIC LIMITED COMPANY

TO THE HONORABLE JAMES M. PECK UNITED STATES BANKRUPTCY JUDGE:

Lehman Brothers Special Financing Inc. ("LBSF") and its affiliated debtors in the

above-referenced chapter 11 cases, including Lehman Brothers Holdings Inc. ("LBHI"), as

debtors and debtors in possession (together, the "Debtors" and, collectively with their non-debtor

affiliates, "Lehman"), file this Motion and respectfully represent:

Background

1. Commencing on September 15, 2008 and periodically thereafter (as applicable, the "<u>Commencement Date</u>"), LBHI and certain of its subsidiaries commenced with this Court voluntary cases under chapter 11 of title 11 of the United States Code (the "<u>Bankruptcy Code</u>"). The Debtors' chapter 11 cases have been consolidated for procedural purposes only and are being jointly administered pursuant to Bankruptcy Rule 1015(b). The Debtors are authorized to operate their businesses and manage their properties as debtors in possession pursuant to sections 1107(a) and 1108 of the Bankruptcy Code.

2. On September 17, 2008, the United States Trustee for the Southern District of New York (the "<u>U.S. Trustee</u>") appointed the statutory committee of unsecured creditors pursuant to section 1102 of the Bankruptcy Code (the "<u>Creditors' Committee</u>").

3. On September 19, 2008, a proceeding was commenced under the Securities Investor Protection Act of 1970 with respect to Lehman Brothers Inc. ("<u>LBI</u>"). A trustee appointed under SIPA is administering LBI's estate.

4. On January 19, 2009, the U.S. Trustee appointed Anton R. Valukas as examiner in the above-captioned chapter 11 cases and by order, dated January 20, 2009 [Docket No. 2583] the Court approved the U.S. Trustee's appointment of the Examiner. On March 11, 2010, the Examiner filed its report with the Court [Docket No. 7531].

5. On March 15, 2010, the Debtors filed their joint chapter 11 plan pursuant to section 1121 of the Bankruptcy Code [Docket No. 7572]. On April 14, 2010, the Debtors filed their revised joint chapter 11 plan [Docket No. 8330] and disclosure statement for their revised joint chapter 11 plan pursuant to section 1125 of the Bankruptcy Code [Docket No. 8332].

2

Jurisdiction

6. This Court has subject matter jurisdiction to consider and determine this matter pursuant to 28 U.S.C. § 1334. This is a core proceeding pursuant to 28 U.S.C. § 157(b).

Lehman's Business

7. Prior to the events leading up to these chapter 11 cases, Lehman was the fourth largest investment bank in the United States. For more than 150 years, Lehman had been a leader in the global financial markets by serving the financial needs of corporations, governmental units, institutional clients and individuals worldwide.

8. Additional information regarding the Debtors' businesses, capital structures, and the circumstances leading to the commencement of these chapter 11 cases is contained in the Affidavit of Ian T. Lowitt Pursuant to Rule 1007-2 of the Local Bankruptcy Rules for the Southern District of New York in Support of First-Day Motions and Applications, filed on September 15, 2008 [Docket No. 2].

Relief Requested

9. By this Motion, the Debtors seek approval, pursuant to Bankruptcy Rule 9019, of a settlement that provides for resolution of all disputes among LBSF, LBHI, Saphir Finance Public Limited Company ("<u>Saphir</u>"), BNY Corporate Trustee Services Limited ("<u>BNY</u>"), The Bank of New York Mellon, London Branch, The Bank of New York Mellon (Ireland) Limited, Perpetual Trustee Company Limited ("<u>Perpetual</u>"), and LBHI Estates Limited (the "<u>Parties</u>"), in connection with two particular series of debt securities issued out of the Dante Program (defined below) and litigation currently pending in this Court as well as the Supreme Court of the United Kingdom (the "<u>UK Supreme Court</u>"). The proposed settlement is reflected in two agreements: (i) the Termination and Settlement Agreement, a copy of which is annexed hereto as "<u>Exhibit A</u>," and (ii) the Settlement Payment Deed (together with the Termination and Settlement Agreement, the "<u>Settlement</u>"), a redacted copy of which is annexed hereto as "<u>Exhibit</u> <u>B</u>."¹ The Debtors have determined, in the exercise of their sound business judgment, that the Settlement is in the best interests of the Debtors, their estates, and their creditors.

The Dante Program

10. BNY, as Trustee, entered into a Principal Trust Deed dated October 10, 2002 as amended and restated from time to time, with Dante Finance Public Limited Company ("<u>Dante</u>") pursuant to which a multi-issuer secured obligation program (the "<u>Dante Program</u>") was established. Under the Dante Program, Saphir, a special purpose entity incorporated in Ireland, issued various series of credit-linked synthetic portfolio notes, including (i) the Series 2004-11 AUD 75,000,000 Synthetic Portfolio Notes Due 2011, and (ii) Series 2006-5 AUD 50,000,000 Synthetic Portfolio Notes due 2011 and Extendable Up to 2016 (collectively, the "<u>Notes</u>") (collectively, the agreements underlying the Saphir 2004-11 and 2006-5 transactions are referred to as the "<u>Transaction Documents</u>"). The Notes were secured by various assets and certain secured obligations. The performance of the Notes is linked to credit-default swap transactions between LBSF and Saphir that were executed in connection with the issuance of the Notes (collectively, the "<u>Transactions</u>")

11. Under the Transactions, the bankruptcy filing of any party constitutes an "Event of Default." Ordinarily, absent an Event of Default by LBSF, the Transaction Documents provide that LBSF received payments before Noteholders, *i.e.*, "Swap Counterparty Priority" applies. Upon an Event of Default by LBSF under the Transactions, however, the Transaction Documents provide that LBSF's payment priority would change such that LBSF

¹ The schedules to the Termination and Settlement Agreement and the Settlement Payment Deed are voluminous and, therefore, have not been attached. Certain confidential monetary and related terms of the Settlement have been redacted from the Settlement Payment Deed. An unredacted copy of the Settlement Payment Deed has been provided to the Court and to the attorneys for the Creditors' Committee.

would receive payments, if any, after payments are made to Perpetual, the sole Noteholder, *i.e.*, a change from "Swap Counterparty Priority" to "Noteholder Priority." In addition, the Transaction Documents further modify LBSF's payment priority by not subtracting from the Noteholder's payment the unwind costs incurred by LBSF.

The Dispute

12. Following the Commencement Date, the Transactions were terminated and Perpetual, the 100% owner of the Notes, sought to redeem the Notes and liquidate the collateral that secures them. Disputes arose between LBSF and Perpetual regarding, *inter alia*, the priority of payments that should be applied by BNY under the Transaction Documents when liquidating the collateral and distributing the proceeds between Perpetual and LBSF.

The English Proceedings

13. On May 13, 2009, Perpetual filed a claim in the High Court of Justice Chancery Division, Royal Courts of Justice, in London (the "<u>English Proceedings</u>"), seeking an order requiring BNY to pay it in accordance with Noteholder Priority. LBSF chose to join the English Proceedings to ensure its rights were protected, although it preferred to litigate these issues in the Bankruptcy Court. The main issue before the High Court was whether the English common law "anti-deprivation principle" would invalidate the shift to Noteholder Priority after the bankruptcy filing. No issues of U.S. law were considered by the High Court.

14. Following the hearing, the High Court ruled in favor of Perpetual. LBSF filed a notice of appeal of the High Court's judgment on August 17, 2009. On November 6, 2009, the English Court of Appeal issued a judgment affirming the previous judgment of the High Court. On March 31, 2010, the UK Supreme Court agreed to hear LBSF's appeal against the Court of Appeal's judgment (the "<u>UK Appeal</u>"). A hearing is scheduled for March 2011.

5

The Adversary Proceeding

15. On May 20, 2009, LBSF filed a complaint in this Court naming BNY as defendant, Adv. No. 09-01242 (JMP) (the "<u>Adversary Proceeding</u>"), seeking a declaratory judgment that (i) provisions in the Transaction Documents purporting to modify LBSF's right to receive priority of payments solely because of its chapter 11 filing are unenforceable "*ipso facto*" clauses, and that, as a result, LBSF is entitled to payment under "Swap Counterparty Priority," and (ii) any attempt to enforce the *ipso facto* clauses would violate the automatic stay.

16. On June 10, 2009, LBSF filed its motion for summary judgment. On June 22, 2009, BNY filed a motion to dismiss the Adversary Proceeding. LBSF opposed the motion to dismiss. On August 11, 2009, the Court denied the motion to dismiss, and thereafter, pursuant to a briefing schedule ordered by the Court, BNY filed a cross motion for summary judgment. On January 25, 2010, this Court granted LBSF's motion for summary judgment in its Memorandum Decision. *Lehman Bros. Special Fin. Inc. v. BNY Corporate Tr. Servs. Ltd. (In re Lehman Bros. Holdings Inc.)*, 422 B.R. 407 (Bankr. S.D.N.Y. 2010). In the Memorandum Decision, this Court held that the *ipso facto* prohibitions of §§ 365(e)(1) and 541(c)(1)(B) of the Bankruptcy Code invalidated the shift to Noteholder Priority, which this court found did not occur prior to LBSF's bankruptcy filing, and held in the alternative that the payment priority shift would constitute an invalid *ipso facto* clause even if it was automatically triggered upon LBHI's earlier bankruptcy filing. *Id.* 418-19.

17. On July 19, 2010, this Court entered an interlocutory order memorializing its Memorandum Decision in favor of LBSF (the "<u>Summary Judgment Order</u>"). On September 20, 2010, the United States District Court for the Southern District of New York (the "<u>SDNY</u> <u>District Court</u>") granted BNY's motion for leave to appeal the Summary Judgment Order (the "<u>US Appeal</u>").

6

18. From time to time over the course of the 18 months following the commencement of the Adversary Proceeding, LBSF and Perpetual have engaged in settlement discussions. When LBSF and Perpetual reached an agreement in principle, LBSF (jointly with the Creditors' Committee) requested that the SDNY District Court stay the US Appeal. On November 19, 2010, Judge Colleen McMahon issued a 90 day stay of the US Appeal. By letter dated November 24, 2010, LBSF has requested that the SDNY District Court remand the US Appeal to this Court so that this Court can consider approval of the Settlement.²

The Settlement

19. The Parties have entered into the Termination and Settlement Agreement

and LBSF, Perpetual, and the Settlement Payment Agent have entered into the Settlement

Payment Deed, both of which are subject to approval by this Court.

20. The salient terms of the Termination and Settlement Agreement are as

follows:³

- the Notes will be redeemed;
- all claims under all of the Transaction Documents (insofar as they relate to the Notes and the Swap Transactions) will be terminated, and the Parties will exchange a mutual release of all claims related to the Transactions; provided, however, that nothing in the Termination and Settlement Agreement shall alter, impair, improve or affect the rights of any holders of notes in the Dante Program (other than the Notes) and such rights (if any) of all such holders of notes (and BNY, to the extent it acts as trustee for the holders of such notes) are reserved in full.

 $^{^2}$ If the remand of the US Appeal is delayed or denied, the parties to the Settlement will confer as to how best to proceed.

³ To the extent that there is any inconsistency between the following summaries and the terms of the Termination and Settlement Agreement and the Settlement Payment Deed, the Termination and Settlement Agreement and the Settlement Payment Deed, as applicable, shall control.

- LBHI, as the Credit Support Provider,⁴ will be relieved of any liability relating to the Transaction Documents;
- the collateral securing the Notes and the Transactions will be sold;
- LBHI Estates Ltd., a non-debtor affiliate of LBSF, will act as the Disposal Agent to sell the collateral;
- following payment of prior ranking costs and expenses, Saphir will pay all moneys payable to LBSF and Perpetual to a Settlement Payment Agent, who will distribute the funds in accordance with the Settlement Payment Deed, to Perpetual and LBSF; and
- following the Effective Date, but prior to receipt of the settlement payment by LBSF, the Adversary Proceeding and the US Appeal will be dismissed, and subsequently, the English Proceedings and the UK Appeal will be dismissed.
- 21. The Termination and Settlement Agreement is conditioned upon
 - a stay of the US Appeal (which, as stated above, was granted by the SDNY District Court);
 - Perpetual obtaining advice from the Supreme Court of New South Wales, Australia, pursuant to the Trustee Act 1925 (NSW), that Perpetual's entry into and performance of its obligations under the Settlement are a proper exercise of its powers and duties, as trustee for noteholders; and
 - the Debtors' obtaining approval of the Settlement from this Court pursuant to Bankruptcy Rule 9019.
- 22. The principal terms of the Settlement Payment Deed are as follows:
 - within one business day of receipt of the payment from Saphir pursuant to the Termination and Settlement Agreement, the Settlement Payment Agent will make a payment to LBSF, and hold the remainder of the moneys until a later date agreed to by the parties;
 - on the agreed payment date, the Settlement Payment Agent will pay a final sum to Perpetual;

⁴ Capitalized terms not otherwise defined herein shall have the meaning ascribed to them in the Termination and Settlement Agreement and the Settlement Payment Deed.

- both Perpetual and LBSF agree to indemnify the Settlement Payment Agent against any liabilities arising from (i) the performance of its obligations under the Settlement Payment Deed, (ii) the Settlement Payment Agent entering into or participating in the Settlement Payment Deed, or (iii) any court proceedings, dispute or arbitration relating to the Settlement Payment Deed, other than liability due to breach of the Settlement Payment Deed, or fraud, negligence, or misconduct;
- the Settlement Payment Deed is subject to the laws of New South Wales, and each party irrevocably agrees that the courts of New South Wales shall have the exclusive jurisdiction to hear and determine any disputes or proceedings relating to the Settlement Payment Deed; and
- each party will bear its own costs and expenses relating to the Settlement Payment Deed. LBSF will pay Perpetual's costs (in the sum of £280,320) relating to a part of the English Proceedings.

The Controlling Legal Standard

23. Bankruptcy Rule 9019(a) provides that, "[o]n motion by the [debtor-in-

possession] and after notice and a hearing, the court may approve a compromise or settlement." Fed R. Bankr. P. 9019(a). This rule empowers bankruptcy courts to approve compromises "if they are in the best interest of the estate." *Vaughn v. Drexel Burnham Lambert Group, Inc. (In re Drexel Burnham Lambert Group, Inc.)*, 134 B.R. 499, 505 (Bankr. S.D.N.Y. 1991); *see also Protective Comm. for Indep. Stockholders of TMT Trailer Ferry, Inc. v. Anderson*, 390 U.S. 414, 424 (1968); *Fisher v. Pereira (In re 47-49 Charles St., Inc.)*, 209 B.R. 618, 620 (S.D.N.Y. 1997); *In re Ionosphere Clubs, Inc.*, 156 B.R. 414, 426 (S.D.N.Y. 1993), *aff*^{*}*d*, 17 F.3d 600 (2d Cir. 1994). Indeed, courts have long considered compromises to be "a normal part of the process of reorganization." *TMT Trailer Ferry*, 390 U.S. at 424 (quoting *Case v. Los Angeles Lumber Prods. Co.*, 308 U.S. 106, 130 (1939).

24. Bankruptcy courts have applied the following factors in determining whether a settlement should be approved: (i) the probability of success in litigation, with due consideration for the uncertainty in fact and law; (ii) the complexity and likely duration of the litigation and any attendant expense, inconvenience, and delay; (iii) the proportion of creditors who do not object to, or who affirmatively support, the proposed settlement; and (iv) the extent to which the settlement is truly the product of arm's-length bargaining and not the product of fraud or collusion. *See Protective Comm. for Indep. Stockholders of TMT Trailer Ferry, Inc.,* 390 U.S. at 424; *In re Ashford Hotels, Ltd.*, 226 B.R. 797, 804 (Bankr. S.D.N.Y. 1998); *In re Best Prods. Co.*, 168 B.R. at 50.

25. The decision to approve a particular compromise lies within the sound discretion of the bankruptcy court. *Nellis v. Shugrue*, 165 B.R. 115, 123 (S.D.N.Y. 1994). The settlement need not result in the best possible outcome for the debtor, but must not fall beneath the lowest point in the range of reasonableness." *In re Drexel Burnham Lambert Group*, 134 B.R. at 505; *see also Cosoff v. Rodman (In re W.T. Grant Co.)*, 699 F.2d 599, 608 (2d Cir. 1983); *In re Spielfogel*, 211 B.R. 133, 144 (Bankr. E.D.N.Y. 1997). Additionally, a court may exercise its discretion "in light of the general public policy favoring settlements." *In re Hibbard Brown* & *Co., Inc.*, 217 B.R. 41, 46 (Bankr. S.D.N.Y. 1998). However, the analysis must focus on the question of whether a particular compromise is "fair and equitable, and in the best interest of the estate." *In re Best Prods. Co.*, 165 B.R. 35, 50 (Bankr. S.D.N.Y. 1994) (internal citations omitted).

26. While a court must "evaluate ... all ... factors relevant to a fair and full assessment of the wisdom of the proposed compromise," *Anderson*, 390 U.S. at 424-25, a court need not conduct a "mini-trial" of the merits of the claims being settled, *W.T. Grant Co.*, 699 F.2d at 608, or conduct a full independent investigation. *Drexel Burnham Lambert Group*, 134 B.R. at 496. "[T]he bankruptcy judge does not have to decide the numerous questions of law and

fact.... The court need only canvass the settlement to determine whether it is within the accepted range of reasonableness." *Nellis*, 165 B.R. at 123 (internal citations omitted).

27. The court may give weight to the "informed judgments of the … debtorin-possession and their counsel that a compromise is fair and equitable, and consider the competency and experience of counsel who support the compromise." *Drexel Burnham Lambert Group*, 134 B.R. at 505 (internal citations omitted); *see also In re Purofied Down Prods. Corp.*, 150 B.R. 519, 522 (S.D.N.Y. 1993); *accord In re Ashford Hotels Ltd.*, 226 B.R. 797, 802 (Bankr. S.D.N.Y. 1998) ("Significantly, that test does not contemplate that I substitute my judgment for the Trustee's, but only that I test his choice for reasonableness.… If the Trustee chooses one of two reasonable choices, I must approve that choice, even if, all things being equal, I would have selected the other.").

28. Significantly, there is no requirement that "the value of the compromise … be dollar-for-dollar the equivalent of the claim." *Ionosphere Clubs, Inc.*, 156 B.R. at 427. Instead, "there is no reason, at least in theory, why a satisfactory settlement could not amount to a hundredth or even a thousandth part of a single percent of the potential recovery." *Id.* at 427-28 (quoting *City of Detroit v. Grinnell Corp.*, 495 F.2d 448 (2d Cir. 1974).

The Settlement Meets the Legal Standard Established Under Rule 9019 and Is in the Bests Interests of the Debtors' Estates

29. The Settlement will clearly benefit the Debtors and their creditors. First, the Settlement will result in a substantial payment to LBSF's estate. LBSF has determined, in the exercise of its business judgment, that such payment will adequately compensate the estate for the value of the Transactions. Second, the Settlement will avoid future disputes and litigation concerning the Transactions because the parties have agreed to release one another from claims thereunder.

30. The Settlement also enables the Debtors to avoid the risks and costs of litigation, including the risks of the US Appeal. Although the Debtors believe that the Court's Summary Judgment Order is grounded in sound bankruptcy law and should be affirmed, the results of litigation are never certain and further litigation (including the potential for further appeals) necessarily will be time consuming and costly to the Debtors' estates. Additionally, if BNY were to be successful in the US Appeal (in whole or in part), there could be a negative impact on the Debtors' efforts to recover value from other similarly-situated counterparties. Finally, the Settlement avoids the need to reconcile the potentially inconsistent decisions of the courts in the United States and the United Kingdom.

31. Thus, the Settlement will allow LBSF to capture the value of the Transactions for its estate, while avoiding the costs associated with continuation of the pending litigation. The Debtors therefore submit that the Settlement falls well above the lowest point in the range of reasonableness. For these reasons, the Court should approve the Settlement pursuant to Bankruptcy Rule 9019.

32. The Creditors' Committee has been involved in the discussion and negotiation of the Settlement, and has been fully informed of and consented to the terms of the Termination and Settlement Agreement and the Settlement Payment Deed.

<u>Notice</u>

33. No trustee has been appointed in these chapter 11 cases. The Debtors have served notice of this Motion on (i) the U.S. Trustee; (ii) the attorneys for the Creditors' Committee; (iii) the Securities and Exchange Commission; (iv) the Internal Revenue Service; (v) the United States Attorney for the Southern District of New York; and (vi) all other parties entitled to notice in accordance with the procedures set forth in the second amended order

12

entered on June 17, 2010 governing case management and administrative procedures for these

cases [Docket No. 9635]. The Debtors submit that no other or further notice need be provided.

34. No previous request for the relief sought herein has been made by the

Debtors to this or any other court.

WHEREFORE the Debtors respectfully request that the Court grant the relief

requested herein and such other and further relief as it deems just and proper.

Dated: November 24, 2010 New York, New York

> <u>/s/ Jacqueline Marcus</u> Ralph I. Miller Peter Gruenberger Jacqueline Marcus

WEIL, GOTSHAL & MANGES LLP 767 Fifth Avenue New York, New York 10153 Telephone: (212) 310-8000 Facsimile: (212) 310-8007

Attorneys for Debtors and Debtors in Possession

Exhibit A

Termination and Settlement Agreement

EXECUTION VERSION

____ NOVEMBER 2010

TERMINATION AND SETTLEMENT AGREEMENT in respect of the Swap Transactions and the related Series 2004-11 AUD 75,000,000 Synthetic Portfolio Notes due 2011 and Series 2006-05 AUD 50,000,000 Synthetic Portfolio Notes due 2011 each issued by Saphir Finance Public Limited Company

between

LEHMAN BROTHERS SPECIAL FINANCING INC. ("LBSF")

SAPHIR FINANCE PUBLIC LIMITED COMPANY (the "Issuer")

BNY CORPORATE TRUSTEE SERVICES LIMITED (formerly J.P. Morgan Corporate Trustee Services Limited) (the "Trustee")

THE BANK OF NEW YORK MELLON, LONDON BRANCH (as successor to JPMorgan Chase Bank, N.A., London Branch) (the "Issuing and Paying Agent" and "Custodian")

THE BANK OF NEW YORK MELLON (IRELAND) LIMITED (formerly BNY Financial Services plc as successor to J.P. Morgan Bank (Ireland) plc) (the "Paying Agent")

PERPETUAL TRUSTEE COMPANY LIMITED (the "Noteholder")

LEHMAN BROTHERS HOLDINGS INC. (the "Credit Support Provider")

> LBHI ESTATES LTD (the "Disposal Agent")

WEIL, GOTSHAL & MANGES

One South Place London EC2M 2WG Tel: +44 (0) 20 7903 1000 Fax: +44 (0) 20 7903 0990

www.weil.com

TABLE OF CONTENTS

1	DEFINI	TIONS AND INTERPRETATION	4
2	CONDI	FIONS PRECEDENT AND EFFECTIVE DATE	11
3	WITHD	RAWAL OF PROCEEDINGS	12
4	TERMI	NATION AND RELEASE	14
5	SETTLE	EMENT SUMS	17
6	PAYME	INTS	18
7	ACCOU	NT DETAILS	20
8	NO SET	2-OFF	20
9	REPRES	SENTATIONS AND WARRANTIES	21
10	AUTHO	RISATION AND RELEASE OF SECURITY	22
11	HOLD H	IARMLESS	23
12	ENTIRE	E AGREEMENT	24
13	CONFIL	DENTIALITY	24
14	NOTICE	ES	26
15	COUNT	ERPARTS	28
16	RIGHTS	S OF THIRD PARTIES	28
17	GENER	AL	28
18	GOVER	NING LAW AND JURISDICTION	29
SCHE	DULE 1	2004-11 Confirmation	30
SCHE	DULE 2	2006-05 Confirmation	31
SCHE	DULE 3	Series Costs	32
SCHE	DULE 4	2004-11 Guarantee	33
SCHE	DULE 5	2006-05 Guarantee	34
SCHE	DULE 6	Sale Procedures	35
SCHE	DULE 7	Form of Extraordinary Resolution	38
EXEC	UTION		39

THIS AGREEMENT is made as a deed on _____ November 2010 (this "**Deed**") between the following parties:

- (1) LEHMAN BROTHERS SPECIAL FINANCING INC. a corporation formed in the State of Delaware, United States of America and whose principal place of business is at 1271 Avenue of the Americas, 40th Floor, New York, NY 10020, United States of America ("LBSF");
- (2) **SAPHIR FINANCE PUBLIC LIMITED COMPANY** a public limited company incorporated in Ireland whose registered office is at AIB International Centre, International Financial Services Centre, Dublin 1, Ireland (the "Issuer");
- (3) BNY CORPORATE TRUSTEE SERVICES LIMITED (formerly J.P. Morgan Corporate Trustee Services Limited) in its capacity as trustee under the Trust Deeds (as defined below) (the "Trustee");
- (4) THE BANK OF NEW YORK MELLON, LONDON BRANCH (as successor to JPMorgan Chase Bank, N.A., London Branch) in its capacity as issuing and paying agent and custodian under the Trust Deeds (the "Issuing and Paying Agent" and the "Custodian", respectively);
- (5) THE BANK OF NEW YORK MELLON (IRELAND) LIMITED (formerly BNY Financial Services plc as successor to J.P. Morgan Bank (Ireland) plc) in its capacity as the paying agent under the Trust Deeds (the "Paying Agent" and together with the Issuing and Paying Agent and the Custodian, the "Agents");
- (6) **PERPETUAL TRUSTEE COMPANY LIMITED** a limited company incorporated in the Commonwealth of Australia with company number ABN 42 000 001 007 whose registered office is at Level 12 Angel Place, 123 Pitt Street Sydney, New South Wales, Australia (the "**Noteholder**");
- (7) LEHMAN BROTHERS HOLDINGS INC. a corporation formed in the State of Delaware, United States of America and whose principal place of business is at 1271 Avenue of the Americas, 40th Floor, New York, NY 10020, United States of America (the "Credit Support Provider"); and
- (8) LBHI ESTATES LTD a private company incorporated in England and Wales whose registered office is at c/o Puxon Murray LLP, 1 Royal Exchange Avenue, London EC3V 3LT, UK in its capacity as disposal agent pursuant to Clause 4.1 (*Sale of Collateral*) (the "Disposal Agent"),
- each a "Party" and together the "Parties".

WHEREAS

- (A) Commencing on 15 September 2008 and thereafter, Lehman Brothers Holdings Inc. and certain of its affiliates, including LBSF, each filed voluntary petitions for relief under chapter 11 of title 11 of the United States Code (the "**Bankruptcy Code**" and the "**Bankruptcy Cases**").
- (B) On 16 December 2008, the court having jurisdiction over the Bankruptcy Cases (the "Bankruptcy Court") entered an order pursuant to Sections 105 and 365 of the

Bankruptcy Code to establish procedures for the settlement or assumption and assignment of prepetition derivative contracts.

- (C) Pursuant to an order of the High Court of Justice in England and Wales dated 3 April 2007 with effect from 19 May 2007 (the "Court Order") all rights and obligations of JPMorgan Chase Bank, N.A., London Branch in any capacity in relation to all documents entered into in connection with the Programme (as defined below) (including all the existing series of notes) were transferred to The Bank of New York (now known as The Bank of New York Mellon), in the manner and to the extent provided in the Court Order.
- **(D)** Dante Finance Public Limited Company ("Dante") and the Trustee are parties to a principal trust deed dated 10 October 2002, as amended and restated from time to time, to which the Issuer has acceded pursuant to: (i) a deed of accession dated 9 April 2003 (in respect of the 2004-11 Notes) (the "2004-11 Accession Deed"); and a deed of accession dated 22 July 2005 (in respect of the 2006-05 Notes) (the "2006-05 Accession Deed" and, together with the 2004-11 Accession Deed, the "Accession Deeds"), establishing a programme (the "Programme") for the issuance from time to time of secured notes (the principal trust deed dated 10 October 2002, as amended and restated from time to time, as acceded to by the Issuer pursuant to the Accession Deeds, the "Principal Trust Deed"). Pursuant to clause 2.7 of the Principal Trust Deed, the Issuer has authorised and issued: (a) its Series 2004-11 AUD 75,000,000 Synthetic Portfolio Notes due 2011 (the "2004-11 Notes"); and (b) its Series 2006-05 Synthetic Portfolio Notes due 2011 extendable up to 2016 (the "2006-05 Notes" and, together with the 2004-11 Notes, the "Notes") each as constituted and secured by the relevant Supplemental Trust Deed.
- (E) LBSF and the Issuer entered into two swap transactions (described more particularly in Recitals (G) and (H) below) governed by an ISDA Master Agreement and Schedule to the ISDA Master Agreement dated as of 10 October 2002 as amended and restated (the "Master Agreement") between LBSF and Dante to which the Issuer has acceded under the Accession Deeds.
- (F) LBSF and the Issuer also entered into a credit support annex to the Master Agreement in connection with each of the 2004-11 Notes and the 2006-05 Notes (together, the "Credit Support Annexes") and a guarantee of LBSF's obligations in respect of each of the Swap Transactions was provided by the Credit Support Provider pursuant to the Credit Support Documents.
- (G) LBSF and the Issuer entered into a credit derivative transaction pursuant to the Master Agreement relating to the 2004-11 Notes with a trade date of 9 December 2004, an effective date of 10 December 2004 and a scheduled termination date of 10 December 2011 (the "2004-11 Swap Transaction") as evidenced by a confirmation dated 10 December 2004, a copy of which is attached in Schedule 1 (2004-11 Confirmation) (the "2004-11 Confirmation").
- (H) LBSF and the Issuer also entered into a credit derivative transaction pursuant to the Master Agreement relating to the 2006-05 Notes with reference number 538204L, an effective date of 17 March 2006 and a scheduled termination date of 17 March 2011 (the "2006-05 Swap Transaction" and, together with the 2004-11 Swap Transaction, the "Swap Transactions") as evidenced by a confirmation dated 17 March 2006, a

copy of which is attached in Schedule 2 (2006-05 Confirmation) (the "2006-05 Confirmation" and, together with the 2004-11 Confirmation, the "Confirmations").

- (I) In connection with the Swap Transactions and the Notes, the Noteholder, LBSF and the Trustee are parties to an action before: (a) the High Court of Justice Chancery Division claim number HC09C01612 (the "UK High Court Proceedings"); (b) the Court of Appeal with appeal number A3/2009/1794 (the "Court of Appeal Proceedings"); and (c) the Supreme Court of the United Kingdom with appeal number UKSC 2009/0221 (the "UK Supreme Court Proceedings" and, together with both of the UK High Court Proceedings and the Court of Appeal Proceedings").
- (J) Furthermore, and also in connection with the Swap Transactions and the Notes, LBSF, the Committee and the Trustee are parties to (a) an adversary proceeding before the United States Bankruptcy Court Southern District of New York, Adv. Proc. No. 09-01242 (JMP) (the "US Bankruptcy Court Proceedings"); and (b) a bankruptcy appeal before the United States District Court Southern District of New York, Case No. M47 (the "District Court Proceedings" and, together with the US Bankruptcy Court Proceedings, the "US Proceedings").
- (K) LBSF and the Noteholder have jointly instructed the Settlement Payment Agent pursuant to the Settlement Payment Deed (as defined below) (the terms of which are confidential between the parties thereto), whereby the Settlement Payment Agent has agreed, among other things, to receive the Termination and Redemption Payments otherwise due to each of LBSF and to the Noteholder from the Issuer in connection with the sale of the Securities Collateral, the redemption of the Notes and the termination of the Swap Transactions.
- (L) Pursuant to two letters, each dated 1 December 2008 headed "Form of Early Termination Notice", the Issuer designated 1 December 2008 as the Early Termination Date in respect of each of the Swap Transactions.
- (M) Notwithstanding the terms of the Swap Documents, LBSF and the Issuer wish to acknowledge the termination of the Swap Transactions, to terminate the Swap Documents and the Trust Deeds and wish to settle fully and finally all matters arising in respect of the UK Proceedings and the US Proceedings (all with effect from and including the Actual Payment Date) and to confirm the terms of the settlement of the Swap Transactions, in consideration of the promises made herein and, more particularly, of the payment by the Issuer to the Settlement Payment Agent of the Termination and Redemption Payments (each as defined below).
- (N) The Issuer is entering into this Deed on the direction of the Noteholder pursuant to an extraordinary resolution substantially in the form attached at Schedule 7 (*Form of Extraordinary Resolution*) (the "**Extraordinary Resolution**"), and with the agreement of LBSF.
- (O) The Trustee is entering into this Deed at the request of LBSF and on the direction of the Noteholder acting pursuant to the Extraordinary Resolution.

IT IS AGREED as follows

1 DEFINITIONS AND INTERPRETATION

1.1 In this Deed:

"Accession Deeds" has the meaning given to such term in Recital (D).

"Actual Payment Date" means the date on which the payments referred to in Clause 6.2 (*Payments*) are paid by or on behalf of the Issuer in accordance with the terms of this Deed (and, if paid on more than one day, then on the day on which the last such payment is made).

"Affiliates" means (a) in the case of a Party, other than the Noteholder, LBSF, the Disposal Agent and the Credit Support Provider, a subsidiary or associated company of that Party or entity in any way related to that Party; (b) in the case of the Noteholder, any Related Body Corporate; and (c) in the case of LBSF, the Disposal Agent and the Credit Support Provider:

- (i) save with respect to the Credit Support Provider, Lehman Brothers Holdings Inc.;
- (ii) a subsidiary or associated entity of Lehman Brothers Holdings Inc. subject, from time to time, to U.S. chapter 11 bankruptcy proceedings (each such subsidiary or associated entity, a "**Related Entity**"); or
- (iii) any other entity under the direct or indirect control or management of either LBSF, Lehman Brothers Holdings Inc., or a Related Entity.

"AUD" means the lawful currency of the Commonwealth of Australia from time to time.

"Australian Court" means the Supreme Court of New South Wales, Australia.

"Australian Court Confirmation" has the meaning given to such term in Clause 2.1 (*Australian Court Advice*).

"Bankruptcy Cases" has the meaning given to such term in Recital (A).

"Bankruptcy Code" has the meaning given to such term in Recital (A).

"Bankruptcy Court" has the meaning given to such term in Recital (B).

"**Business Day**" means a day other than a Saturday, Sunday or recognised public holiday in England and Wales, the United States of America or Australia, on which the majority of large deposit banks in London are open for business.

"Cash Amount" means the 2004-11 Cash Amount and the 2006-05 Cash Amount.

"**Committee**" means the court-sanctioned official committee of unsecured creditors representing the creditors of LBSF (amongst other affiliated entities).

"Conditions" means the terms and conditions of the Notes.

"**Conditions Precedent**" means (a) (unless the District Court Proceedings have already been stayed as at the date of this Deed) the receipt by LBSF of a Fax Receipt in respect of the US Stay Confirmation (as sent to each of the Trustee and the Noteholder in accordance with

Clause 2.3 (*Stay of US Proceedings*)); (b) the receipt by LBSF of the Australian Court Confirmation; and (c) the receipt by LBSF of a Fax Receipt in respect of the notification given pursuant to Clause 2.2 (*US Bankruptcy Court Authority*) in respect of the LBSF Authorisation Confirmation (sent to each of the Trustee and Noteholder pursuant to Clause 2.2 (*US Bankruptcy Court Authority*)), each in accordance with the terms of this Deed.

"Confidential Information" has the meaning given to such term in Clause 13.1 (*Confidentiality*).

"Confirmations" has the meaning given to such term in Recital (H).

"Court Order" has the meaning given to such term in Recital (C).

"Court of Appeal Proceedings" has the meaning given to such term in Recital (I).

"Credit Support Annexes" has the meaning given to such term in Recital (F).

"Credit Support Documents" means the 2004-11 Guarantee and the 2006-05 Guarantee.

"**Credit Support Provider**" means Lehman Brothers Holdings Inc. a corporation formed in the State of Delaware, United States of America and whose principal place of business is located at 1271 Avenue of the Americas, 40th Floor, New York, NY 10020, United States of America.

"CSA Cash Collateral" means the 2004-11 CSA Cash Collateral and the 2006-05 CSA Cash Collateral.

"Dante" has the meaning given to such term in Recital (D).

"District Court Proceedings" has the meaning given to such term in Recital (J).

"Effective Date" has the meaning given to such term in Clause 2.4 (Effective Date).

"Euro", "€" or "EUR" means the lawful currency and composite monetary unit of the European Union.

"Extraordinary Resolution" has the meaning set out in Recital (N).

"Fax Receipt" has the meaning given to such term in Clause 14.8 (*Notices*).

"GBP" or "£" means the lawful currency of the United Kingdom, from time to time.

"Information" has the meaning given to such term in Clause 11.1 (Hold Harmless).

"**Issuer Contingency**" means the amount listed as being the Issuer Contingency in the schedule of Series Costs which is provided to LBSF and the Noteholder pursuant to Clause 6.1 (*Payments*).

"Issuer Contingency Account" has the meaning given to such term in Clause 6.8(a) (*Payments*)

"**Issuer Excess Contingency**" means the amount (if any) of the Issuer Contingency which, as at the Issuer Final Determination Date, has not been actually incurred by the Issuer or its advisors.

"Issuer Final Determination Date" means the date on which the Issuer notifies the Trustee and the Noteholder that it has finally determined that no further costs are likely to be incurred in relation to the Notes. The Issuer agrees that it will review at regular intervals following the Actual Payment Date whether it is appropriate to make such determination and will use reasonable efforts to make the notification expeditiously after such determination. The Issuer Final Determination Date shall occur no later than the Third Anniversary unless the Issuer provides written notice to the Trustee and to the Noteholder prior to the Third Anniversary of circumstances which, in the reasonable opinion of the Issuer may give rise to further costs being incurred in relation to the Notes after the Third Anniversary, in which case the Issuer Final Determination Date shall occur no later than the date six (6) years from the date of this Deed.

"LBSF Authorisation Confirmation" has the meaning given to such term in Clause 2.2 (US *Bankruptcy Court Authority*).

"Mahogany I Notes" has the meaning given to such term in Clause 11.5 (Hold Harmless).

"Mahogany II Notes" has the meaning given to such term in Clause 11.5 (Hold Harmless).

"Master Agreement" has the meaning given to such term in Recital (E).

"**Mortgaged Property**" means both the 2004-11 Mortgaged Property and the 2006-05 Mortgaged Property.

"Motion to Stay" has the meaning given to such term in Clause 2.3 (*Stay of US Proceedings*).

"Notes" has the meaning given to such term in Recital (D).

"Party" and "Parties" have the meaning given to such terms at the beginning of this Deed.

"**Payment Instruction Notice**" means the notice served by either of LBSF or the Noteholder pursuant to Clause 3.3 (*Payment Instruction Notice*) which confirms to the recipients thereof that each of the US Proceedings and the UK Proceedings have been withdrawn and the Sale Amounts have been received and instructs the Issuer to expeditiously make the payments set out in Clause 6.2 (*Payments*).

"Principal Trust Deed" has the meaning given to such term in Recital (D).

"Programme" has the meaning given to such term in Recital (D).

"**Related Body Corporate**" has the meaning given to that such term in section 50 of the Corporations Act 2001 (Cth) (as understood in accordance with the laws of New South Wales, Australia).

"Related Entity" has the meaning given to such term in the definition of Affiliates.

"**Related Party**" and "**Related Parties**" have the meanings given to such terms in Clause 13.1 (*Confidentiality*).

"**Released Party**" has the meaning given to such term in Clause 4.3(a) (*Release*) and "**Released Parties**" shall be construed accordingly.

"**Reserved Rights**" has the meaning given to such term in Clause 4.5 (*Outstanding Transactions*).

"Sale Amounts" means both of the 2004-11 Sale Amount and the 2006-05 Sale Amount.

"Sale Procedures" mean the sale procedures as set out in Schedule 6 (*Sale Procedures*) which shall apply separately in respect of the sale of each of the 2004-11 Securities Collateral and the 2006-05 Securities Collateral.

"Securities Collateral" means both of the 2004-11 Securities Collateral and the 2006-05 Securities Collateral.

"Series Costs" means the costs properly incurred in connection with the Notes or this Deed or any of the transactions contemplated by this Deed, by the Trustee, and any relevant agent under the Notes, the Issuer and their respective advisers.

"Settlement Payment Agent" means Henry Davis York lawyers, 44 Martin Place, Sydney, NSW 2000 Australia.

"**Settlement Payment Deed**" means the settlement and payment deed dated on or about the date of this Deed made amongst LBSF, the Noteholder and the Settlement Payment Agent (the terms of which are confidential between the parties thereto).

"**Supplemental Trust Deeds**" means both of the 2004-11 STDD Agreement and the 2006-05 STDD Agreement.

"Swap Documents" means the Confirmations and, so far as incorporated into the Confirmations, the Master Agreement, together with the Credit Support Annexes and the Credit Support Documents.

"Swap Transactions" has the meaning given to such term in Recital (H).

"Termination and Redemption Payments" means both of the 2004-11 Termination and Redemption Payments and the 2006-05 Termination and Redemption Payments.

"Third Anniversary" means the date which falls three (3) years from the date of this Deed.

"**Transaction Documents**" means the Trust Deeds, the Swap Documents, the Notes and any other transaction document to the extent that such transaction document relates to the Notes or to the Swap Transactions.

"Trust Deeds" means both of the 2004-11 Trust Deed and the 2006-05 Trust Deed.

"**Trustee Contingency**" means the amount listed as being the Trustee Contingency in the schedule of Series Costs which is provided to LBSF and the Noteholder pursuant to Clause 6.1 (*Payments*).

"Trustee Excess Contingency" means the amount (if any) of the Trustee Contingency which, as at the Trustee Final Determination Date, has not been actually incurred by the Trustee or any relevant Agent or their respective advisers under the Notes.

"**Trustee Final Determination Date**" means the date on which the Trustee notifies the Issuer and the Noteholder that it has finally determined that no further costs are likely to be incurred in relation to the Notes. The Trustee agrees that it will review at regular intervals following the Actual Payment Date whether it is appropriate to make such determination and will use reasonable efforts to make the notification expeditiously after such determination. The Trustee Final Determination Date shall occur no later than the Third Anniversary unless the Trustee provides written notice to the Noteholder prior to the Third Anniversary of circumstances which, in the reasonable opinion of the party giving the notice may give rise to further costs being incurred in relation to the Notes after the Third Anniversary, in which case the Trustee Final Determination Date shall occur no later than the date 6 years from the date of this Deed.

"**Unpaid Series Costs**" means both of the 2004-11 Unpaid Series Costs and the 2006-05 Unpaid Series Costs.

"**UK High Court Withdrawal Confirmation**" means the confirmation provided in accordance with Clause 14 (*Notices*) by the Noteholder to each of LBSF and the Trustee confirming that the UK High Court Proceedings have been dismissed pursuant to Clause 3.2(b) (*Withdrawal of the UK Proceedings*).

"UK High Court Proceedings" has the meaning given to such term in Recital (I).

"UK Proceedings" has the meaning given to such term in Recital (I).

"UK Supreme Court Proceedings" has the meaning given to such term in Recital (I).

"UK Supreme Court Withdrawal Confirmation" means the confirmation provided in accordance with Clause 14 (*Notices*) by LBSF to each of the Noteholder and Trustee confirming that the notice of appeal before the UK Supreme Court has been withdrawn pursuant to Clause 3.2(a) (*Withdrawal of the UK Proceedings*).

"US Appeal Withdrawal Confirmation" means the confirmation provided in accordance with Clause 14 (*Notices*) by the Trustee to each of the Noteholder and LBSF confirming that the District Court Proceedings have been dismissed pursuant to Clause 3.1(b) (*Dismissal of the US Proceedings*).

"US Bankruptcy Court Proceedings" has the meaning given to such term in Recital (J).

"US Bankruptcy Court Withdrawal Confirmation" means the confirmation provided in accordance with Clause 14 (*Notices*) by LBSF to each of the Noteholder and the Trustee confirming that the US Bankruptcy Court Proceedings have been dismissed pursuant to Clause 3.1(a) (*Dismissal of the US Proceedings*).

"US Proceedings" has the meaning given to such term in Recital (J).

"US Stay Confirmation" means the confirmation provided in accordance with Clause 14 (*Notices*) by LBSF to each of the Trustee and the Noteholder confirming that the District Court Proceedings have been stayed pursuant to Clause 2.3(a) (*Stay of the US Proceedings*).

"Winning Bid" and "Winning Bidder" have the meanings given to such terms in paragraph 5 of the Sale Procedures.

"**Withdrawal Date**" means the later of (a) the withdrawal of the UK High Court Proceedings and (b) the dismissal of the District Court Proceedings.

"2004-11 Accession Deed" has the meaning given to such term in Recital (D).

"**2004-11 Cash Amount**" means the cash amount standing to the credit of those of the Issuer's bank accounts that relate to the 2004-11 Notes, including, without limitation, the 2004-11 CSA Cash Collateral.

"2004-11 Confirmation" has the meaning given to such term in Recital (G).

"2004-11 CSA Cash Collateral" means AUD 43,884.20 (plus any interest accrued thereon between 12 November 2010 and the Actual Payment Date), being the cash amount standing to the credit of those of the Issuer's bank accounts that relate to the Credit Support Annex relating to the 2004-11 Swap Transaction.

"**2004-11 Guarantee**" means the guarantee provided by the Credit Support Provider in favour of the Issuer relating to the 2004-11 Swap Transaction, a copy of which is attached in Schedule 4 (*2004-11 Guarantee*).

"2004-11 Mortgaged Property" has the meaning given to such term in the 2004-11 STDD Agreement.

"2004-11 Notes" has the meaning given to such term in Recital (D).

"**2004-11 Sale Amount**" means the aggregate amount(s) received from the Winning Bidder(s) in respect of the Winning Bid(s) for the 2004-11 Securities Collateral pursuant to the Sale Procedures.

"**2004-11 Securities Collateral**" means AUD 75,000,000 floating rate notes issued by Australia and New Zealand Banking Group Limited (ISIN: XS0208078732) maturing 10 December 2011.

"2004-11 STDD Agreement" means the supplemental trust deed and drawdown agreement dated 10 December 2004 between, inter alios, the Trustee, the Issuer and LBSF constituting the 2004-11 Notes.

"2004-11 Swap Transaction" has the meaning given to such term in Recital (G).

"**2004-11 Termination and Redemption Payments**" means the 2004-11 Cash Amount plus the 2004-11 Sale Amount minus the 2004-11 Unpaid Series Costs.

"**2004-11 Trust Deed**" means the 2004-11 STDD Agreement and, so far as incorporated into the 2004-11 STDD Agreement, the Principal Trust Deed.

"**2004-11 Unpaid Series Costs**" means all Series Costs excluding those which have been paid as at the date of this Deed attributed to the 2004-11 Notes.

"2006-05 Accession Deed" has the meaning given to such term in Recital (D).

"2006-05 Cash Amount" means the cash amount standing to the credit of those of the Issuer's bank accounts that relate to the 2006-05 Notes, including, without limitation, the 2006-5 CSA Cash Collateral.

"2006-05 Confirmation" has the meaning given to such term in Recital (H).

"2006-05 CSA Cash Collateral" means AUD 301,732.51 (plus any interest accrued thereon between 12 November 2010 and the Actual Payment Date), being the cash amount standing to the credit of those of the Issuer's bank accounts that relate to the Credit Support Annex relating to the 2006-5 Swap Transaction.

"**2006-05 Guarantee**" means the guarantee provided by the Credit Support Provider in favour of the Issuer relating to the 2006-05 Swap Transaction, a copy of which is attached in Schedule 5 (*2006-05 Guarantee*).

"2006-05 Mortgaged Property" has the meaning given to such term in the 2006-05 STDD Agreement.

"2006-05 Notes" has the meaning given to such term in Recital (D).

"**2006-05 Sale Amount**" means the aggregate amount(s) received from the Winning Bidder(s) in respect of the Winning Bid(s) for the 2006-05 Securities Collateral pursuant to the Sale Procedures.

"**2006-05 Securities Collateral**" means AUD 50,000,000 floating rate notes issued by The Royal Bank of Scotland PLC (ISIN: XS0247983058) maturing in 2011 extendable up to 17 March 2016.

"2006-05 STDD Agreement" means the supplemental trust deed and drawdown agreement dated 17 March 2006 between, inter alios, the Trustee, the Issuer and LBSF constituting the 2006-05 Notes.

"2006-05 Swap Transaction" has the meaning given to such term in Recital (H).

"**2006-05 Termination and Redemption Payments**" means the 2006-05 Cash Amount plus the 2006-05 Sale Amount minus the 2006-05 Unpaid Series Costs.

"2006-05 Trust Deed" means the 2006-05 STDD Agreement and, so far as incorporated into the 2006-05 STDD Agreement, the Principal Trust Deed

"2006-05 Unpaid Series Costs" means all Series Costs excluding those which have been paid as at the date of this Deed attributed to the 2006-05 Notes.

1.2 Terms used but not defined herein shall have the meanings given to them in the Swap Documents and/or the Trust Deeds.

1.3 Wording denoting the singular shall include the plural and vice versa; and the words "include" and "including" shall be construed without limitation.

1.4 All references in this Deed to any corporate body, company or firm shall include the successors in title or assigns from time to time.

1.5 The headings and sub-headings in this Deed are inserted for convenience only and do not affect its interpretation.

1.6 Unless the context otherwise requires, a reference to a Clause or Schedule is a reference to a clause of, or schedule to, this Deed.

1.7 For the avoidance of doubt, references to the Trust Deeds in this Deed (including, without limitation, any references to their respective termination), shall be construed as referring to each of: (a) the 2004-11 STDD Agreement and, only so far as incorporated therein, the Principal Trust Deed; and (b) the 2006-05 STDD Agreement and, only so far as incorporated therein, the Principal Trust Deed. Nothing in this Deed shall be construed to mean that the Principal Trust Deed is terminated with respect to any other series in the Programme.

2 CONDITIONS PRECEDENT AND EFFECTIVE DATE

2.1 Australian Court Advice

The Noteholder shall forthwith upon execution by the Parties of this Deed, apply to the Australian Court for judicial advice pursuant to the Trustee Act 1925 (NSW) (to be construed in accordance with the laws of New South Wales, Australia), substantially to the effect that its entry into and performance of its obligations under this Deed and the Settlement Payment Deed are a proper exercise of its powers and duties (such advice to be in form and substance satisfactory to the Noteholder acting reasonably). If the Noteholder has obtained the advice so applied for from the Australian Court in accordance with this Clause 2.1, it shall expeditiously notify each of the Trustee and LBSF that the necessary court advice has been obtained (the "Australian Court Confirmation"). If the Noteholder has obtained the advice and, acting reasonably, judges it not to be in form and substance satisfactory to the Noteholder, it shall expeditiously notify (by fax and email) each of the Trustee and LBSF of the reasons why the court advice is not in form and substance satisfactory to it.

2.2 US Bankruptcy Court Authority

LBSF shall forthwith upon the execution by the Parties of this Deed, seek an order authorising LBSF to enter into and perform its obligations under this Deed and the Settlement Payment Deed (a final order will constitute the "LBSF Authorisation Confirmation"). Once LBSF has obtained the LBSF Authorisation Confirmation, it shall expeditiously notify each of the Trustee and the Noteholder that the necessary court order has been obtained and provide a copy of such order.

2.3 Stay of US Proceedings

(a) Unless the District Court Proceedings have already been stayed in which case the US Stay Confirmation will be deemed to have been sent and received, LBSF shall forthwith upon execution by the Parties of this Deed, move to stay the District Court Proceedings for a period of 90 days (the "Motion to Stay") and the Trustee, acting on instructions of the Noteholder which are hereby given, agrees not to object or oppose the Motion to Stay. Once LBSF has confirmation from the United States District Court Southern District of New York that the Motion to Stay has been granted, LBSF will expeditiously send the US Stay Confirmation.

(b) Motions for any extension to the stay of the District Court Proceedings shall be made by either the Trustee (upon written request by the Noteholder (with the agreement of LBSF)) or by LBSF (with the agreement of the Noteholder).

2.4 Effective Date

- (a) LBSF shall forthwith upon satisfaction of the Conditions Precedent, give notice thereof to each of the Issuer, the Trustee, the Noteholder and the Disposal Agent. If LBSF fails to give such notice within two Business Days after the date on which the Noteholder reasonably believes (based on such information, if any, as may then be available to the Noteholder) the final Condition Precedent to be so satisfied, the Noteholder may give such notice to the Issuer, the Trustee, LBSF and the Disposal Agent, by email and fax (provided that such notice will set out reasonable details of the basis on which the Noteholder considers all Conditions Precedent to have been satisfied).
- (b) Other than the provisions of this Clause 2 (*Conditions Precedent and Effective Date*), and Clauses 1, 9, 11, 12, 13, 14, 15, 16, 17, and 18 which all shall take effect on the date hereof, the remaining provisions of this Deed shall take effect immediately upon the date on which there are successful Fax Receipts in respect of the notices to each Party to whom notice is required to be given in accordance with Clause 2.4(a) above (the "**Effective Date**").

3 WITHDRAWAL OF PROCEEDINGS

3.1 Dismissal of the US Proceedings

- (a) LBSF shall, forthwith upon the Effective Date, expeditiously move jointly with the Trustee to have the US Bankruptcy Court Proceedings dismissed with prejudice. Once LBSF has confirmation from the Bankruptcy Court that the US Bankruptcy Court Proceedings have been dismissed with prejudice, LBSF will expeditiously send the US Bankruptcy Court Withdrawal Confirmation.
- (b) The Trustee shall, forthwith upon the Effective Date, expeditiously move jointly with LBSF to have the District Court Proceedings dismissed with prejudice. Once the Trustee has confirmation from the United States District Court for the Southern District of New York that the District Court Proceedings have been dismissed with prejudice, the Trustee will expeditiously send the US Appeal Withdrawal Confirmation.

3.2 Withdrawal of the UK Proceedings

(a) LBSF shall, with the consent of the Noteholder which is hereby provided, forthwith upon sending the US Bankruptcy Court Withdrawal Confirmation, withdraw the notice of appeal before the UK Supreme Court. Pursuant to Rule 34 of the Rules of the Supreme Court 2009, LBSF shall notify the UK Supreme Court of the parties' consent to withdraw the notice of appeal (with no order as to costs), without an oral hearing, subject to the permission of the UK Supreme Court. Once LBSF has confirmation from the UK Supreme Court that the notice of appeal before the UK Supreme Court has been withdrawn, LBSF shall expeditiously send the UK Supreme Court Withdrawal Confirmation.

(b) The Noteholder shall, with the consent of each of LBSF and the Trustee which is hereby provided, forthwith upon receipt of the US Bankruptcy Court Withdrawal Confirmation, make an application to the English High Court for the UK High Court Proceedings to be dismissed with no order as to costs. Once the Noteholder has confirmation from the UK High Court that the UK High Court Proceedings have been dismissed, the Noteholder will expeditiously send the UK High Court Withdrawal Confirmation.

3.3 Payment Instruction Notice

- (a) Forthwith upon the occurrence of the last of the following:
 - (i) the receipt by LBSF of a Fax Receipt in respect of the US Bankruptcy Court Withdrawal Confirmation (sent by LBSF to each of the Trustee and the Noteholder pursuant to Clause 3.1(a) (*Dismissal of the US Proceedings*));
 - (ii) the receipt by LBSF of the US Appeal Withdrawal Confirmation (sent by the Trustee to each of LBSF and the Noteholder pursuant to Clause 3.1(b) (*Dismissal of the US Proceedings*));
 - (iii) the receipt by LBSF of a Fax Receipt in respect of the UK Supreme Court Withdrawal Confirmation (sent by LBSF to each of the Trustee and the Noteholder pursuant to Clause 3.2(a) (*Withdrawal of the UK Proceedings*));
 - (iv) the receipt by LBSF of the UK High Court Withdrawal Confirmation (sent by the Noteholder to each of the Trustee and LBSF pursuant to Clause 3.2(b) (*Withdrawal of the UK Proceedings*)); and
 - (v) the receipt by the Issuer of the Sale Amounts (which is to be confirmed by the Trustee to each of LBSF and the Noteholder expeditiously after receipt of the Sale Amounts by the Issuer),

LBSF shall deliver the Payment Instruction Notice to each of the Issuer, the Trustee and the Noteholder by email and fax.

(b) If LBSF fails to deliver the Payment Instruction Notice in accordance with Clause 3.3(a) above, within two Business Days after the date on which the Noteholder reasonably believes (based on such information, if any, as may then be available to the Noteholder) the final of the items listed in Clause 3.3(a)(i) to (v) above to have occurred, the Noteholder may serve to each of the Issuer, the Trustee and LBSF, the Payment Instruction Notice by email and fax (provided that such notice will set out details of the basis on which the Noteholder believes that the items listed in Clause 3.3(a)(i) to (v) have been satisfied).

3.4 Opinion of District Court

Notwithstanding anything to the contrary in this Deed or otherwise and notwithstanding that the Effective Date may have already occurred, to the extent that the District Court issues an opinion on the merits of the appeal in the District Court Proceedings prior to the delivery of the Payment Instruction Notice, this Deed shall terminate and cancel in all respects and all of the rights and obligations created under and/or pursuant to it shall be terminated and released immediately without any further notice or action from any Party save for this Clause 3.4 and

the following Clauses: 1 (*Definitions and Interpretation*), 9.8 (*Representations and Warranties*), 11 (*Hold Harmless*), 13 (*Confidentiality*) and 18 (*Governing Law and Jurisdiction*), the provisions of which shall remain binding on all Parties, on the basis that no Party shall have any claim against any of the other Parties under or in connection with this Deed except in relation to a breach by any Party of any of such surviving Clauses. For the avoidance of doubt and without limitation to the generality of the forgoing, none of the releases contained within this Deed shall be effected and each of the Parties shall reserve its rights to challenge or appeal the opinion issued by the District Court in respect of the District Court Proceedings.

4 TERMINATION AND RELEASE

4.1 Sale of Collateral

- (a) The Issuer hereby appoints LBHI Estates Ltd to act as its Disposal Agent solely in connection with the sale of the Securities Collateral in accordance with the Sale Procedures, and LBHI Estates Ltd hereby accepts such appointment. LBHI Estates Ltd's duties as Disposal Agent shall only extend to arranging a sale of the Securities Collateral in accordance with the Sale Procedures, and LBHI Estates Ltd shall have no liability to any other Party in connection with the sale of the Securities Collateral. Notwithstanding the terms of the Trust Deeds or any other documents in relation to the Notes, none of LBHI Estates Ltd nor any of its Affiliates shall receive any fees, remuneration, commission or any other payment with respect to LBHI Estates Ltd's role as Disposal Agent. In addition, none of LBHI Estates Ltd nor any of its Affiliates shall directly or indirectly purchase the Securities Collateral.
- (b) All Parties hereby agree to the Sale Procedures as set out in Schedule 6 (*Sale Procedures*). The Parties also agree that the Disposal Agent shall initiate the Sale Procedures:
 - (i) in respect of the 2004-11 Securities Collateral, two (2) Business Days after the Effective Date; and
 - (ii) in respect of the 2006-05 Securities Collateral, five (5) Business Days after the Effective Date,

or on such other Business Day as may be agreed from time to time between the Noteholder, LBSF and the Disposal Agent. If at any time (at least one Business Day in advance of the proposed sale date in accordance with either Clause 4.1(b)(i) or (ii) above), the Noteholder is of the view (acting reasonably) that the market conditions are illiquid or otherwise not conducive to achieving the best price for the sale of the relevant Securities Collateral, the Noteholder may delay the Sale Procedures for a period of up to (but no more than) six (6) weeks and designate such day(s) within that period to conduct the Sale Procedures (subject to agreement with LBSF (such agreement not to be unreasonably withheld or delayed), having given at least one Business Day's notice of the proposed date(s) to the Disposal Agent).

(c) A Winning Bid submitted to the Disposal Agent in accordance with the Sale Procedures shall be deemed to comply with any obligation arising (where applicable) under rule 11.2 of the Financial Service Authority's Conduct of Business Sourcebook (Best Execution).

4.2 The Termination and Redemption Payments and the CSA Cash Collateral

- (a) Notwithstanding the terms of the Master Agreement and the Confirmations:
 - (i) the payment of the Termination and Redemption Payments to the Settlement Payment Agent (acting on behalf of the Noteholder and LBSF) shall be deemed to constitute and satisfy all sums payable to LBSF in connection with any of the Swap Documents, the Notes or the Trust Deeds and in respect of the termination of the Swap Transactions;
 - (ii) the payment of the Termination and Redemption Payments to the Settlement Payment Agent (acting on behalf of the Noteholder and LBSF) and the payment of the Issuer Excess Contingency (if any) and the Noteholder Excess Contingency (if any) shall be deemed to constitute and satisfy all sums payable to the Noteholder in connection with any of the Swap Documents, the Notes or the Trust Deeds and in respect of the redemption of the Notes; and
 - (iii) no further amounts shall be due or payable or collateral shall be deliverable by LBSF, the Credit Support Provider, the Issuer or the Trustee in connection with either: (X) the Swap Documents and their termination; or (Y) the Notes and their redemption.
- (b) The Issuer (acting in this regard on the instructions of the Noteholder and at the request of LBSF) shall pay the Termination and Redemption Payments to the Settlement Payment Agent in accordance with Clauses 6 (*Payments*), 7 (*Account Details*) and 8 (*No Set-off*).
- (c) LBSF hereby confirms, and the Issuer and Trustee hereby agree (which agreement of the Issuer and the Trustee is solely on the instruction of the Noteholder), that as at the date of this Deed, the CSA Cash Collateral is LBSF's Credit Support Balance pursuant to the Credit Support Annexes.
- (d) The Issuer shall pay the CSA Cash Collateral (as part of the Termination and Redemption Payments) to the Settlement Payment Agent (acting on behalf of the Noteholder and LBSF) in accordance with Clauses 6 (*Payments*), 7 (*Account Details*) and 8 (*No Set-off*). The Issuer covenants that it will comply with paragraph 3(a) of the Credit Support Annexes in paying the CSA Cash Collateral to the Settlement Payment Agent, save to the extent that this paragraph was not complied with when the relevant CSA Cash Collateral was delivered to the Issuer.
- (e) LBSF confirms that payment of the CSA Cash Collateral as set out in Clause 4.2(d) (*The Termination and Redemption Payments and the CSA Cash Collateral*) above shall be in satisfaction of the Issuer's obligation to transfer the Return Amount to LBSF under the terms of the Credit Support Annexes.

4.3 Release

(a) Notwithstanding the terms of the Transaction Documents, in consideration of each other Party's execution of this Deed and conditional upon and with effect from payment to the Settlement Payment Agent of the Termination and Redemption Payments referred to in Clause 6.2 (*Payments*) (without set-off, withholding or

counterclaim) in accordance with the terms of this Deed, and subject to Clause 11.4 (Hold Harmless), each Party hereby generally and irrevocably releases, discharges and acquits each other Party together with its respective Affiliates and current and former agents, servants, officers, directors, shareholders, employees, subsidiaries, divisions, branches, units, parents, attorneys, successors, predecessors, heirs, personal representatives and assigns (each of the foregoing a "Released Party") from: (i) any and all obligations to each other Party with respect to the Transaction Documents and their respective rights against each other thereunder are cancelled (including, without limitation, any rights, liabilities or obligations of the Parties with respect to payments or other obligations due and payable or due to be performed on or prior to the Actual Payment Date); and (ii) without prejudice to (i) above, all manners of action, causes of action, judgments, executions, debts, demands, rights, damages, costs, expenses and claims of every kind, nature and character whatsoever whether in law or in equity, whether based on contract (including, without limitation, quasi contract or estoppel), regulation. tort (including. without limitation. intentional statute. torts. misrepresentation, defamation, breaches of alleged fiduciary duty, recklessness, gross negligence, or negligence) or otherwise, accrued or unaccrued, known or unknown, matured or unmatured, liquidated or unliquidated, certain or contingent, that such releasing Party ever had or claimed to have or now has or claims to have presently or at any future date against each Released Party arising under or related to the Transaction Documents, their negotiation, execution, performance, any payments thereunder, any breaches thereof, or their termination.

- (b) For the avoidance of doubt, nothing herein has the effect of releasing any claim or potential claim which any of: (i) Mahogany Capital Limited ACN 107 882 753 (in Liquidation); (ii) the holders of the Mahogany I Notes; or (iii) the holders of the Mahogany II Notes, may have solely in connection with the sale or marketing of the Notes, the Mahogany I Notes, or the Mahogany II Notes, provided always, that nothing contained in this Deed shall be construed or interpreted as a recognition or acknowledgement by any Party of any of the following: (A) that any Party has any liability in connection with the sale or marketing of the Notes; or (C) that any valid claim exists (or has ever existed) against any Party in connection with the sale or marketing of the Notes. For the avoidance of doubt, nothing in this Clause 4.3(b) shall limit or affect the releases of the Trustee provided in this Deed.
- (c) Subject to Clause 4.3(d) below, each Party agrees that (subject to Clause 11.4 (*Hold Harmless*) (i) the Trust Deeds; (ii) the Swap Documents; and (iii) any other Transaction Document (but without prejudice to the ongoing validity of any such Transaction Document to the extent it relates to any other series in the Programme), and all rights and obligations of each Party thereunder, shall be terminated with effect from the Actual Payment Date, and subject to that LBSF hereby agrees that it shall not seek to challenge the validity of the termination of the Swap Transactions, or other Transaction Documents or the redemption of or any payments made in respect of the Notes or seek any other relief in connection with the Transaction Documents other than the termination and the related payments provided herein.
- (d) Solely in respect of each of the Issuer, the Trustee and the Noteholder, the terminations provided for in Clause 4.3(c) above shall not apply to the Trust Deeds and the Transaction Documents in respect of the right and entitlement of the

Noteholder, the Trustee and the Issuer to any Trustee Contingency or Issuer Contingency; in such regard only, the Trust Deeds and the Transaction Documents shall terminate upon the Issuer Final Determination Date (with respect to the Issuer Contingency) and the Trustee Final Determination Date (with respect to the Trustee Contingency). To the extent that any such portions of the Trust Deeds and/or the other Transactions Documents conflict with, or are inconsistent with, any of the terms of this Deed, the terms of this Deed shall govern; and in no event shall the Trustee or the Issuer be entitled to any fees during the period from the date of this Deed until the date of any application of any amount of Trustee Contingency or Issuer Contingency.

4.4 Release of the Credit Support Provider

For the avoidance of doubt, the Issuer and the Trustee (acting at the express direction of the Noteholder) intend and agree that, with effect from payment to the Settlement Payment Agent of the payments referred to in Clause 6 (*Payments*):

- (a) the Credit Support Provider is released and discharged from further obligation with respect to the Swap Documents;
- (b) their respective rights against the Credit Support Provider with respect to the Credit Support Documents are cancelled; and
- (c) the Credit Support Documents granted by the Credit Support Provider are hereby cancelled.

4.5 Outstanding Transactions

The obligations and releases contained in this Deed (and the termination of any agreements or documents) relate only to the Notes, the Swap Transactions and the agreements and other arrangements related thereto (including the Transaction Documents) and do not affect any Party's rights and obligations with respect to any other series in the Programme or any other agreement or arrangement entered into between the Parties. For the avoidance of doubt, (i) nothing in this Clause 4 (Termination and Release) shall release a Party from its obligations under this Deed and (ii) nothing in this Deed shall alter, impair, improve or affect the rights of any holders of notes in the Programme (other than the Notes) to assert the validity of the provisions of the documents relating to such notes (or to defend any challenge to the validity of those provisions) in any proceedings brought in the United States Bankruptcy Court or in any other Court and such rights (if any) of all such holders of notes (and the Trustee, to the extent it acts as trustee for the holders of such notes) are reserved in full (the "Reserved **Rights**") provided always that nothing in this Deed shall constitute any sort of acknowledgement or recognition on behalf of LBSF or the Debtors or any other Party that any of the holders of such notes can make any valid claim or defence in respect of such Reserved Rights.

5 SETTLEMENT SUMS

5.1 Each Party acknowledges that the amounts set out in this Deed as payable to, or to the Settlement Payment Agent on behalf of, such Party accurately and completely represent the sums to be paid to each of them pursuant to Clause 4.2 (*The Termination and Redemption Payments and the CSA Cash Collateral*) and Clause 6 (*Payments*).

5.2 In particular:

- (a) LSBF recognises that the payment of the Termination and Redemption Payments to the Settlement Payment Agent (acting on behalf of the Noteholder and LBSF), shall be in satisfaction of any amounts owed or due to LBSF in connection with the termination of the Swap Transactions and any entitlement LBSF may have pursuant to the Transaction Documents or as a beneficiary of any trust established pursuant to the Transaction Documents, and no other amounts shall be due or payable in connection therewith to LBSF; and
- (b) the Noteholder recognises that the payment of the Termination and Redemption Payments to the Settlement Payment Agent (acting on behalf of the Noteholder and LBSF) and the payment of the Trustee Excess Contingency (if any) and the Issuer Excess Contingency (if any), shall be in satisfaction of any amounts owed or due to the Noteholder in connection with the redemption of the Notes and any entitlement the Noteholder may have pursuant to the Transaction Documents or as a beneficiary of any trust established pursuant to the Transaction Documents, and no other amounts shall be due or payable in connection therewith to the Noteholder.

6 **PAYMENTS**

6.1 The Parties agree that on or prior to the Actual Payment Date, the Trustee will notify the Noteholder and LBSF of the amounts owing in respect of the Series Costs for each of the 2004-11 Notes and the 2006-05 Notes, each in the form set out in Schedule 3 (*Series Costs*). The Noteholder shall deliver the Notes to the Issuing and Paying Agent upon the earlier to occur of (i) the Noteholder receiving confirmation from the Settlement Payment Agent of its receipt of the Termination and Redemption Payments; and (ii) three Business Days after the Actual Payment Date, as follows:

- (a) in the case of the 2004-11 Notes, the Noteholder shall arrange for delivery of the 2004-11 Notes to the Issuer to Euroclear account number 14037 by way of surrender for nil consideration; and
- (b) in the case of the 2006-05 Notes, the Noteholder shall direct the custodian holding the 2006-05 Notes to deliver the 2006-05 Notes by recorded international courier to the Issuing and Paying Agent at the address set out in Clause 14.4 (*Notices*) (marked for the attention of Richard Kerry) by way of surrender for nil consideration,

and upon receipt of the relevant Notes, the Issuing and Paying Agent shall promptly cancel the Notes.

6.2 Following the receipt by the Issuer of the Payment Instruction Notice, the Parties agree (notwithstanding any other provision to the contrary contained in the Conditions) that the Issuer shall apply the Cash Amount and the Sale Amounts to make, or procure to be made, the following payments and provisions (and, immediately before making such payments, all rights, title and interest of the Issuer in, to and under the Cash Amount and the Sale Amounts (other than the amounts paid to the Trustee in respect of Unpaid Series Costs which are to be paid to the Issuer) shall, notwithstanding Clause 6.7 below, cease and be of no further force or effect):

(a) the Unpaid Series Costs to the Trustee;

(b) provision for the Issuer Contingency and the Trustee Contingency; and

(c) the Termination and Redemption Payments to the Settlement Payment Agent.

6.3 The Issuer shall (to the extent that it has it within its control to do so) use reasonable efforts to, and make any necessary direction to the Agents to, procure that: (i) the payments set out in Clause 6.2 (*Payments*) are made expeditiously following receipt of the Payment Instruction Notice; and (ii) the settlement directions in respect of such payments are given as close in time to each other as is reasonably practicable.

6.4 The Trustee shall, to the extent within its powers, pursuant to the direction of the Noteholder and at the request of LBSF, take any actions or make any directions to the Agents reasonably required to procure that the payments set out in Clause 6.2 (*Payments*) are made expeditiously following the receipt of the Payment Instruction Notice pursuant to Clause 3.3 (*Payment Instruction Notice*).

6.5 The Noteholder and/or LBSF shall provide to the Trustee and the Issuer (and/or the Agents) such information as the Trustee and the Issuer (and/or the Agents) may reasonably request in order to facilitate such payments.

6.6 Neither the Issuer, the Trustee nor any of the Agents shall have any obligation to make such payments on any particular date and nor shall any of them have any liability to any other Party or any other person in connection with any such payment not being made on any particular date provided that the above shall not release the Issuer of its obligations under Clause 6.3 (*Payments*) or the Trustee from its obligations under Clause 6.4 (*Payments*).

6.7 The Trustee agrees that the funds in respect of the Trustee Contingency shall be maintained in the Issuer's account held by the Custodian under the control of the Trustee, and that to the extent there is any Trustee Excess Contingency as at the Trustee Final Determination Date, it will notify the Noteholder in writing of the Trustee Excess Contingency and make payment of the Trustee Excess Contingency to the Noteholder in accordance with Clause 7 (*Account Details*). The Trustee agrees that it will use reasonable efforts to procure that the notification and payment of the Trustee Excess Contingency is made expeditiously following the Trustee Final Determination Date.

- **6.8** The Issuer agrees that:
- (a) it will request the Custodian to open a new account in the name of the Issuer (under the control of the Trustee) to hold the Issuer Contingency (the "Issuer Contingency Account") to be distributed in accordance with this Deed;
- (b) it will notify the Custodian, the Trustee and the Noteholder in writing of any costs that have been actually and properly incurred by the Issuer or its advisors and for which the Issuer seeks reimbursement from the sums held in the Issuer Contingency Account; and
- (c) if there is any Issuer Excess Contingency as at the Issuer Final Determination Date, it will notify the Custodian, the Trustee and the Noteholder in writing of the Issuer Excess Contingency.
- 6.9 The Custodian is hereby instructed by the Issuer and the Noteholder that:

- (a) upon receipt of a notice from the Issuer pursuant to Clause 6.8(b) above, the Custodian shall make payment to the Issuer in respect of costs that have been actually and properly incurred from the sums held at that time in the Issuer Contingency Account; and
- (b) upon receipt of a notice from the Issuer pursuant to Clause 6.8(c) above, the Custodian shall make payment of the Issuer Excess Contingency to the Noteholder in accordance with Clause 7 (*Account Details*).

6.10 The Issuer shall use reasonable efforts to procure that the notification and payment of the Issuer Excess Contingency is made expeditiously following the Issuer Final Determination Date.

6.11 The Parties acknowledge that to the extent that any of the Series Costs are in a currency other than AUD the Trustee shall be entitled to convert those costs into AUD in accordance with clause 9.13 (*Currency Conversion*) of the Principal Trust Deed.

7 ACCOUNT DETAILS

(a) Payment of the Termination and Redemption Payments to the Settlement Payment Agent shall be made to the following account:

Henry Davis York Trust AccountWestpac Banking Corporation, SydneySWIFT:WPAC AU2SBSB:032-000A/C:300-114Reference:3117305;

(b) Payment of the Issuer Excess Contingency (if any) and the Trustee Excess Contingency (if any) shall be made to the Noteholder to such account as the Noteholder shall specify.

8 NO SET-OFF

The Termination and Redemption Payments payable by the Issuer to the Settlement Payment Agent (acting on behalf of the Noteholder and LBSF) under this Deed shall be calculated and made:

- (a) without (and free and clear of any deduction for) any set-off, netting, withholding, combination of accounts or retention in respect of any present or future monies, debts or liabilities owing to the Issuer or incurred by LBSF or any subsidiaries, associated companies or entities which are in any way related to LBSF, or any other party; and
- (b) regardless of any right of equity or counterclaim the Issuer may have against LBSF or any subsidiaries, associated companies or any entities in any way related to LBSF, or any other party.

9 **REPRESENTATIONS AND WARRANTIES**

9.1 On the Effective Date and on the Actual Payment Date, each Party, except for the Agents and the Trustee, represents and warrants to the other Parties (except to the Agents) that:

- (a) the execution and delivery by it of, and the performance by it of its obligations under, this Deed are within its corporate powers and have been duly authorised by all necessary action and, in the case of LBSF only, this Deed is being entered into in accordance with and pursuant to the LBSF Authorisation Confirmation;
- (b) the execution and delivery by it of, and the performance by it of its obligations under, this Deed will not contravene any regulations, provision of applicable law or, the constitutional documents of it, or any agreement or other instrument binding upon itself or affecting it or any of its assets, or any judgment, order or decree of any governmental body, agency or court having jurisdiction over it, and no consent, approval, authorisation or order of, or qualification with, any governmental body or agency is required for the performance by it of its obligations under this Deed and it is not in breach of or in default under any agreement or instrument to which it is a party or which is binding on it or any of its assets or revenues, which breach or default is likely to be material in the context of entering into this Deed; and
- (c) when executed and delivered by it and the other Parties, this Deed will constitute valid and legally binding agreements of it, enforceable against it in accordance with its terms.

9.2 Each Party, except for the Agents and the Trustee, further represents and warrants to the other Parties (except the Agents) that:

- (a) it is not relying upon any statements, understandings, representations, expectations, or agreements other than those expressly set forth in this Deed;
- (b) it has had the opportunity to be represented and advised by legal counsel in connection with this Deed, which it enters voluntarily and of its own choice and not under coercion or duress;
- (c) it has made its own investigation of the facts and is relying upon its own knowledge and the advice of its counsel; and
- (d) it has no expectation that any of the other Parties will disclose facts material to the Swap Documents, the Trust Deed or this Deed.

9.3 The Issuer represents and warrants to each of the other Parties (except for the Agents) that:

- (a) it is the sole owner of all rights (other than those of LBSF, the Trustee and the Credit Support Provider) under the Swap Documents;
- (b) it is duly authorised to compromise all of such rights; and
- (c) it has not transferred or assigned any of such rights prior to the date of this Deed, except by way of security pursuant to the Trust Deeds.

9.4 LBSF represents and warrants to each of the other Parties (except for the Agents) that it has not assigned or transferred by way of security or otherwise (and will not seek to do so) any of its rights, title, liabilities, obligations or interest under or in connection with the Transaction Documents.

9.5 The Noteholder represents that at the date hereof it: (a) is the holder of 100 per cent. in principal amount outstanding of the Notes; (b) has not transferred or otherwise disposed of its interest under the Notes; (c) in the case of the 2004-11 Notes, the Noteholder has blocked its account with the relevant clearing systems; and (d) in the case of the 2006-05 Notes, the Noteholder has not given (and will not give) any instructions to the custodian holding the 2006-05 Notes to transfer the 2006-05 Notes other than as set out in Clause 6.1 (*Payments*). The Noteholder covenants that it will not give directions for the account to be un-blocked until after all payments have been made in accordance with this Deed.

9.6 The Noteholder represents and warrants, on each date on which, pursuant to the terms of this Deed, any interest is paid on the Notes, that it is resident for the purposes of tax in the Commonwealth of Australia and is not operating in Ireland through a branch or agency with which the interest payments are connected.

9.7 LBSF represents and warrants that, other than the US Proceedings and the UK Proceedings, there are no adversary proceedings or other proceedings brought by LBSF or any of its Affiliates, or to which LBSF or any of its Affiliates is a party, whether or not as part of the Bankruptcy Cases, arising under or related to the Transaction Documents.

9.8 Each of the Parties agrees and stipulates that: (i) it shall not assert, or support an assertion by any person or entity, that the Noteholder is subject to the jurisdiction of any court in the United States with respect to the Transaction Documents or this Deed; and (ii) none of the terms and conditions of this Deed, or the actions of any Party in entering into, complying with, or effecting any of the terms hereof, shall be deemed to be, or be treated as, evidence that the Noteholder is subject to such jurisdiction.

9.9 The Parties agree and stipulate that each Party (other than the Agents) is relying upon the representations and warranties in this Clause 9 (*Representations and Warranties*) in entering into this Deed. Furthermore, the Parties (other than the Agents) agree that these representations and warranties are a material inducement for entering into this Deed.

10 AUTHORISATION AND RELEASE OF SECURITY

10.1 The Trustee, on the express direction of the Noteholder and at the request of LBSF, authorises the Issuer to: (a) enter into this Deed (and carry out all acts necessary to give effect to the terms of this Deed) for the purposes of clauses 2.5 (*Payment After a Default*) and/or 5.4 (*Rights of the Issuer*), in each case, to the extent applicable and 7.1.15 (*Restrictions*) of the Principal Trust Deed; and (b) to make the payments set out in Clause 4.2 (*The Termination and Redemption Payments and the CSA Cash Collateral*) and Clause 6 (*Payments*) in accordance with this Deed for the purposes of clause 7.1.15 (*Restrictions*) of the Principal Trust Deed.

10.2 All amounts paid by or on behalf of the Issuer pursuant to Clause 6.2 (*Payments*) shall be paid free and clear of all charges and security interests created by the Issuer under the Trust Deeds (including in respect of the Termination and Redemption Payments) and the Trustee, on the express direction of the Noteholder, irrevocably releases, discharges and

reassigns to the Issuer any rights relating thereto and all the other amounts assigned to the Trustee pursuant to the Trust Deeds immediately prior to such payments being made to the extent necessary to enable such payments to be so made in accordance with the Trust Deeds.

10.3 The Trustee and the Issuer are entering into this Deed at the request and direction of the Noteholder and at the request of LBSF, and in reliance upon their respective representations. The holdings of the Noteholder have been verified to the Trustee in accordance with the normal practices of the clearing systems. On such basis, the Trustee confirms that it has all necessary power and authority to enter into this Deed.

11 HOLD HARMLESS

11.1 To the extent that a Party (except for the Issuer or the Trustee) has received and relied on any information analysis or explanation (the "Information") from the Trustee or the Issuer (or any of its advisers) in connection with the steps envisaged by this Deed, each Party (except for the Issuer and the Trustee) agrees that (i) such Information is confidential save for (a) information which becomes generally known to the public, (b) information which is required to be disclosed pursuant to any applicable laws or to any competent governmental or statutory authority or pursuant to rules or regulations of any relevant regulatory, administrative or supervisory body (including, without limitation, any relevant stock exchange or securities council), (c) information which is required to be disclosed pursuant to any legal process issued by or commenced in any court or tribunal, (d) disclosure to a Party's Affiliates, subsidiaries or related corporations, (e) disclosure in confidence to its respective professional advisers or insurers, including, in the case of LBSF, the Committee and their advisers, (f) information which is required to be disclosed to the extent necessary to comply with each of the Parties' obligations under Clauses 2 (Conditions Precedent and Effective Date) and 3 (Withdrawal of Proceedings), (g) information which is required to be disclosed for the purposes of enforcing this Deed in accordance with its terms, and (h) disclosure to each Party's respective directors, officers and employees, in each case solely for the purpose of effecting this Deed or as reasonably required for internal compliance or reporting purposes (it being understood that such parties will be informed of the confidential nature of the Information); (ii) the person supplying it shall not owe such Party any duty or liability in respect of the Information; (iii) if a Party relies on such information, it does so at its own risk; and (iv) such Party shall not bring, or threaten to bring, any actions, proceedings or claims against the person supplying it or any of its advisers in any way arising out of, or in connection with, the Information.

11.2 By executing this Deed and pursuant to the Extraordinary Resolution, the Noteholder, in accordance with the Trust Deeds and other transaction documents in respect of the Notes hereby directs and instructs the Trustee and the Issuer to execute this Deed and to be subject to the rights and obligations contained herein that are applicable to them.

11.3 Each of the Parties (other than the Issuer) irrevocably agrees that it will at no time institute or join any person in bringing, instituting or joining, any insolvency or examinership proceedings (whether court-based or otherwise) in relation to the Issuer. Each Party hereby agrees that its recourse against the Issuer in respect of any claim arising out of or in connection with this Deed or the Transaction Documents be limited to the Mortgaged Property (or the proceeds thereof) and if, having realised the Mortgaged Property (or the proceeds thereof), the Issuer has insufficient funds to discharge any amounts arising from such claim then the relevant Party shall have no further claim against the Issuer in respect of

such unpaid amounts and such unpaid amounts shall be deemed discharged in full and extinguished.

11.4 Notwithstanding any other provision of this Deed, nothing in this Deed shall restrict or limit the Trustee or the Issuer from any indemnification claim, or benefiting from any hold harmless or protection provisions, set out in the Trust Deeds, any other transaction document relating to the Notes provided that this Clause 11.4 (*Hold Harmless*) shall not apply to any such indemnification claim, hold harmless or protection provisions provided by LBSF, the Credit Support Provider, the Disposal Agent or any of their respective Affiliates.

11.5 The Noteholder enters into this Deed in the following capacities and no other capacity: (i) the holder of record of the 2004-11 Notes and the 2006-05 Notes; (ii) note trustee and security trustee in relation to the issue of note by Mahogany Capital Limited and known as the Mahogany Notes Series I (the "**Mahogany I Notes**"); (iii) note trustee and security trustee in relation to the issue of notes by Mahogany Capital Limited and known as the Mahogany Notes Series II (the "**Mahogany I Notes**"). The liability of the Noteholder arising under or in connection with this Deed is limited to, and can be enforced against it only to, the extent to which it can be satisfied out of the property and assets held by the Noteholder for the time being and from time to time in its capacity as trustee for the Mahogany I Notes and for the Mahogany II Notes from which the Noteholder is indemnified for the liability. This limitation of liability applies despite any other provision of this Deed and extends to all liabilities and obligations of the Noteholder in any way connected with any representation, warranty, conduct, omission, indemnity, agreement or transaction made in or related to this Deed.

12 ENTIRE AGREEMENT

12.1 This Deed (and, in the case of the Parties to it, the Settlement Payment Deed) constitutes the entire agreement and understanding of the Parties relating to the subject matter hereof and supersedes all prior agreements and understandings relating to the subject matter hereof.

12.2 Each Party acknowledges that, in entering into this Deed, it is not relying on any statement, representation, assurance or warranty of any person (whether a Party to this Deed or not) other than as expressly set out in this Deed.

12.3 Each Party agrees that the only rights and remedies available to it arising out of or in connection with any representation or warranty set out in this Deed shall be damages for breach of contract.

12.4 Nothing in this Clause shall limit or exclude liability for fraud.

13 CONFIDENTIALITY

13.1 Save to the extent necessary to comply with each of the Parties' obligations under Clauses 2 (*Conditions Precedent and Effective Date*) and 3 (*Withdrawal of Proceedings*), each Party undertakes to each other Party that they shall not at any time disclose the contents of this Deed, including the Termination and Redemption Payments ("**Confidential Information**"), to any person except:

- (a) each Party's respective directors and officers, in each case who need to know such information solely for the purpose of effecting this Deed or as reasonably required for internal compliance or reporting purposes;
- (b) accountants, auditors, legal or professional advisers, insurers (each of whom are themselves subject to a duty of confidentiality),

(the entities listed at Clauses 13.1(a) and (b) above each being a "**Related Party**" (and together the "**Related Parties**"), it being understood that such Related Parties will each be informed of the confidential nature of the Confidential Information and will be requested to maintain the confidentiality of such Confidential Information);

- (c) in the case of LBSF, the Committee and their advisers;
- (d) other Parties to this Deed;
- (e) as may be required by law, including but not limited to, as may be required by the United States federal securities and bankruptcy laws;
- (f) as may be required: (i) in connection with the amendment or withdrawal of any proof of claim in the bankruptcy of LBSF or the Credit Support Provider; (ii) by any applicable listing rule or listing law; or (iii) pursuant to the authority of a court, administrative tribunal, arbitration panel, regulatory or supervisory body and/or their duly appointed representatives; or
- (g) to the extent necessary to enforce the terms of this Deed.

13.2 Each Party further undertakes that any communication of Confidential Information shall include a prominent legend to the effect that the Confidential Information shall be treated as confidential and may not be subject to onward distribution.

13.3 Each Party (other than LBSF, the Trustee and the Agents) agrees that, when it enters into a confidentiality agreement with respect to Confidential Information, it shall promptly provide LBSF a copy of said agreement.

13.4 If any Party (other than LBSF, the Trustee and the Agents) or any Related Party (of any Party other than LBSF, the Trustee or the Agents) becomes legally obligated (whether by court or regulatory order or otherwise), other than due to disclosure by such Party or Related Party which is permitted by Clause 13.1 above, to disclose any of the Confidential Information, such Party or such Related Party will promptly endeavour to provide LBSF if permitted by law, with prompt notice of such proposed disclosure so that LBSF may seek a protective order or other appropriate remedy. If such a protective order or other protective remedy is not obtained, such Party or such Related Party, as the case may be, will disclose only that portion of the Confidential Information which is legally required, in the opinion of its own counsel, and such Party or such Related Party, as the case may be, will exercise its reasonable efforts to obtain reliable assurances that confidential treatment will be accorded the Confidential Information.

13.5 For the avoidance of doubt, Confidential Information is not intended to cover information which is already in the public domain (provided that any such information did not enter the public domain by way of breach of this Deed).

13.6 Notwithstanding any other provision of this Clause 13 (*Confidentiality*) none of the Trustee or the Issuer shall be obliged to do, or refrain from doing, anything which would cause it to breach its obligations under the Trust Deeds.

14 NOTICES

Any communication by any Party to another in relation to this Deed may be sent by fax, prepaid post or delivered personally to the address set out below or to such other address as a Party may stipulate, except for any notices required to be sent under Clause 2 (*Conditions Precedent and Effective Date*) or Clause 3 (*Withdrawal of Proceedings*) which shall be sent by email and fax only (with delivery by fax being the effective delivery when delivered in accordance with Clause 14.8 (*Notices*)):

14.1 LBSF

c/o Lehman Brothers Holdings Inc. 1271 Avenue of the Americas, 40th Floor New York, NY 10020 USA Attention: Derivatives Legal Fax: +1 646-285-9701 Email: derivativeslegal-london@lamcollc.co.uk, Andrew.Weber@lamcollc.co.uk, Locke.Mcmurray@lamcollc.com and Adrian.Elliott@lamcollc.co.uk

and

c/o LAMCO Services Ltd 6th Floor 111 Old Broad Street London EC2N 1FP Attention: Derivatives Legal Fax: +44(0) 20 7492 3008

14.2 The Issuer

Saphir Finance Public Limited Company AIB International Centre International Financial Services Centre Dublin 1 Ireland Attention: The Directors Fax: +353 1 874 3050 Email: edwina.c.fogarty@aib.ie and fergus.gillen@mccannfitzgerald.ie

14.3 The Trustee

BNY Corporate Trustee Services Limited One Canada Square London E14 5AL Attention: Corporate Trust Administration Fax: +44 20 7964 2530

Email: dagemea@bnymellon.com

14.4 The Issuing and Paying Agent and the Custodian

The Bank of New York Mellon, London Branch One Canada Square London E14 5AL Attention: Corporate Trust Administration Fax: +44 20 7964 2530

14.5 The Paying Agent

The Bank of New York Mellon (Ireland) Limited Guild House Guild Street, Dublin 2 Ireland Attention: Corporate Trust Administration Fax: +44 20 7964 2530

14.6 The Noteholder

Perpetual Trustee Company Limited Level 12 Angel Place 123 Pitt Street, Sydney New South Wales Australia Attention: Group Executive, Corporate Trust Fax: +61 (2) 8256 1401 Email: <u>chris.green@perpetual.com</u>

14.7 The Disposal Agent

LBHI Estates Ltd c/o LAMCO Services Ltd 6th Floor 111 Old Broad Street London EC2N 1FP Attention: Derivatives Legal Fax: +44(0) 20 7492 3008 Email: <u>derivativeslegal-london@lamcollc.co.uk</u>, <u>Andrew.Weber@lamcollc.co.uk</u>, Locke.Mcmurray@lamcollc.com and <u>Adrian.Elliott@lamcollc.co.uk</u>

14.8 Communications by post will be deemed served two days (but if to another country, seven days) after posting. Communications by fax will be deemed served when transmission has been completed and a transmission confirmation has been obtained by the sender (a "**Fax Receipt**").

14.9 If a Party is notified that its email address or fax number set out above is incorrect, or otherwise becomes aware of such an error, it shall forthwith correct such details by notice via fax and, save for the Agents, email to each other Party.

15 COUNTERPARTS

This Deed may be executed and delivered in counterparts (including by fax transmission or in AdobeTM Portable Document Format (PDF) sent by electronic mail), each of which when executed and delivered shall constitute an original of this Deed, but all of the counterparts shall together constitute the same Deed.

16 RIGHTS OF THIRD PARTIES

16.1 Save as set out in Clause 16.2 (*Rights of Third Parties*), no person other than a Party to this Deed shall have any right (whether under the Contracts (Rights of Third Parties) Act 1999 or otherwise) to enforce or enjoy the benefit of any term of this Deed.

16.2 The provisions of Clause 4.3 (*Release*) is intended to benefit each Released Party, and may be enforced (subject to the remaining terms of this Deed) by the relevant Released Party, as applicable, as if it were a Party.

17 GENERAL

17.1 The provisions of this Deed will be binding upon and inure to the benefit of the Parties and their respective successors and assigns.

17.2 No variation to this Deed shall be effective unless in writing and signed on behalf of each of the Parties. For the avoidance of doubt, no consent of a Released Party shall be required in respect of any variation.

17.3 The waiver by any of the Parties of any breach of this Deed shall not be deemed a waiver of any other prior or subsequent breach of this Deed. No waiver of any right under this Deed shall be effective unless it is in writing and signed by the Party waiving such right.

17.4 If any term or provision in this Deed (other than the obligation to make the payments specified in Clause 6 (*Payments*) and other than Clause 3.4 (*Opinion of District Court*)) shall be held to be invalid, illegal or unenforceable, in whole or in part, in any respect under the law of any jurisdiction, the validity, legality and enforceability of any such provision in every other respect and of the remaining provisions contained in this Deed shall not be in any way impaired thereby, it being intended that all of the rights and privileges of the Parties shall be enforceable to the fullest extent permitted by law. Upon such determination that any term or other provision is invalid, illegal or unenforceable, the Parties shall negotiate in good faith to modify this Deed so as to effect the original intent of the Parties as closely as possible in an acceptable manner so that the transactions contemplated by this Deed are consummated as originally understood to the greatest extent possible.

17.5 Each of the Parties named below confirms that it has irrevocably and unconditionally appointed the person named opposite its name below as its agent to accept service of process on its behalf in England in respect of any proceedings arising out of or in connection with this Deed. Such service shall be deemed to be validly made notwithstanding that such person shall not notify any Party thereof. Nothing in this Clause shall affect the right to serve process in any other manner permitted by law.

Party	Agent
The Issuer	McCann FitzGerald, St Michael's House, 1 George Yard, Lombard

	Street, London EC3V 9DE
LBSF	LAMCO Services Ltd, 6th Floor, 111 Old Broad Street, London EC2N 1FP

18 GOVERNING LAW AND JURISDICTION

18.1 This Deed and any non-contractual obligations arising out of or in connection with it shall be governed by and construed in accordance with English law.

18.2 The Parties irrevocably agree that the courts of England are to have exclusive jurisdiction to settle any dispute which may arise out of or in connection with this Deed. The Parties irrevocably submit to the jurisdiction of those courts and waive any objection to proceedings in any such court on the ground of venue or on the ground that the proceedings have been brought in an inconvenient forum, or on the grounds that the proceedings are subject to LBSF's automatic stay under the Bankruptcy Code or otherwise.

This Deed has been entered into and delivered as a Deed on the date stated at the beginning of this Deed.

EXECUTION

EXECUTED AND DELIVERED AS A DEED BY LEHMAN BROTHERS SPECIAL FINANCING INC. (as "LBSF") By:)))
Signature:)
EXECUTED AND DELIVERED AS A DEED BY LEHMAN BROTHERS HOLDINGS INC. (as the " Credit Support Provider ") By:)))
Signature:)
SIGNED, SEALED AND DELIVERED AS A DEED BY PERPETUAL TRUSTEE COMPANY LIMITED (as the " Noteholder ") by its attorney under Power of Attorney dated)))
Signature of attorney:)
Print name:)
In the presence of: Signature of Witness:))
Name of Witness:)
Address of Witness:)
EXECUTED AND DELIVERED AS A DEED BY BNY CORPORATE TRUSTEE SERVICES LIMITED (as the " Trustee "))))
By:)
Signature:)
By:)
Signature:)

EXECUTED AND DELIVERED AS A DEED BY THE BANK OF NEW YORK MELLON, LONDON BRANCH (as the "Issuing and Paying Agent")) H))
By:)
Signature:)
By:)
Signature:)
EXECUTED AND DELIVERED AS A DEED BY THE BANK OF NEW YORK MELLON, LONDON BRANCH (as the " Custodian ")) H))
By:)
Signature:)
By:)
Signature:)
SIGNED AND DELIVERED AS A DEED for and on behalf of THE BANK OF NEW YORK MELLON (IRELAND) LIMITED (as the "Paying Agent") acting by its duly authorised attorney:)))))
Title: Attorney-in-Fact)
Name:)
In the presence of: Signature of Witness:))
Name of Witness:)
Address of Witness:)
Occupation of Witness:)

SIGNED AND DEL	IVERED AS A DEED BY)
Signature of attorney:)
Name of attorney:)
as attorney for SAPH (as the " Issuer ")	IR PUBLIC LIMITED COMPANY	
In the presence of:)
Signature of Witness:)
Name of Witness:)
Address of Witness:	c/o Riverside One Sir John Rogerson's Quay Dublin 2, Ireland)))
EXECUTED AND I LBHI ESTATES LT (as the " Disposal Age)))
By:)
Signature:)
In the presence of: Signature of Witness:))
Name of Witness:)
Address of Witness:)

Exhibit B

Settlement Payment Deed



Execution version

Settlement Payment Deed

Perpetual Trustee Company Limited ACN 000 001 007

Lehman Brothers Special Financing Inc.

Henry Davis York Lawyers

Settlement Payment Deed	HENRY DAVIS YORK

CONTENTS

KEY	DETAILS	1
BAC	KGROUND	2
TER	MS	3
1	Interpretation	3
2	Appointment of Settlement Payment Agent	6
3	Receipt and disbursement of the Termination and Redemption Payments	6
4	Legal costs of the UK Proceedings	7
5		7
6	Duties of Settlement Payment Agent	8
7	Confidentiality	10
8	Counterparts	13
9	Entire agreement	13
10	Further assurance	13
11	Representations and Warranties	13
12	Governing law	14
13	Legal costs	15
14	Notices	15
15	Waiver	16
16	Severability	17
17	Payment	17
18	Time	18

Settlement Payment Deed		HENRY DAVIS YORK
Schedule 1	Permitted Disclosures	19
Schedule 2	Termination and Settlement Agreement	20
Schedule 3	Confidentiality Protocol	21

Settlement Payment Deed

HENRY DAVIS YORK

KEY DETAILS

1	Date	See Execution on page 20
2	Parties	
	Perpetual	
	Name	Perpetual Trustee Company Limited ABN 42 000 001 007
	Address	Level 12 Angel Place 123 Pitt Street Sydney, New South Wales, Australia
	Attention	Group Executive, Corporate Trust
	Fax	+61 2 8256 1401
	Email	chris.green@perpetual.com.au
	LBSF	
	Name	Lehman Brothers Special Financing Inc.
	Address	1271 Sixth Avenue, 40th Floor, New York, NY 10020, United States of America
	Attention	Locke McMurray
	Fax	+1 646-285-9701
	Email	locke.mcmurray@lamcollc.com
	Settlement Payment	
	Agent	
	Name	Henry Davis York lawyers
	Address	44 Martin Place,
		Sydney, New South Wales 2000 Australia
	Attention	Alex Mufford
	Fax	+612 9947 6999
	Email	alex_mufford@hdy.com.au

BACKGROUND

- A Perpetual, LBSF and BNY Corporate Trustee Services Limited (**BNY**) are parties to an action before:
 - (a) the High Court of Justice Chancery Division claim number HC09C01612 (High Court Proceedings);
 - (b) the Court of Appeal appeal number A3/2009/1794 (**Court of Appeal Proceedings**); and
 - (c) the Supreme Court of the United Kingdom appeal number UKSC
 2009/0221 (Supreme Court Proceedings and together with the High Court
 Proceedings and the Court of Appeal Proceedings, the UK Proceedings).
- B LBSF, the Committee and BNY are parties to:
 - (a) an adversary proceeding in the United States Bankruptcy Court Southern District of New York, Adv. Proc. No. 09-01242 (JMP) (Bankruptcy Court Proceedings); and
 - (b) a bankruptcy appeal in the United States District Court Southern District of New York, Case No. M47 (**District Court Proceedings** and together with the Bankruptcy Court Proceedings, the **US Proceedings**).
- C The UK Proceedings and the US Proceedings concern various issues in dispute between the respective parties in their various capacities arising out of two synthetic credit-linked note transactions under the so-called "Dante" multi-issuer secured obligation programme.
- D Perpetual, BNY and LBSF have agreed to settle fully and finally all matters arising in respect of the two synthetic credit-linked note transactions, the UK Proceedings and the US Proceedings on the terms set out in the Termination and Settlement Agreement between, among others, Perpetual, BNY and LBSF dated on or about the date of this deed.
- E Perpetual and LBSF agree to appoint the Settlement Payment Agent to receive and then pay the Termination and Redemption Payments on the terms and conditions set out in this deed.

Settlement Payment Deed

TERMS

1 Interpretation

1.1 **Definitions**

Terms used but not defined in this deed shall have the meanings given to such terms in the Termination and Settlement Agreement.

The following words have the following meanings in this deed, unless the context requires otherwise.

Australian Court means the Supreme Court of New South Wales.

Business Day means a day that is not a Saturday, Sunday or public holiday in any of Sydney, New South Wales, London or New York.

Confidential Information has the meaning given in clause 7.1.

Confidentiality Windback Date means

Confidentiality Protocol means the protocol contained in Schedule 3.

Disputes means any disputes or claims which may arise out of or in connection with this deed including:

- (a) regarding its existence, validity or termination; and
- (b) any non-contractual obligations or liabilities arising out of or in connection with it.

Key Details means the section of this deed headed Key Details.

LBSF Settlement Amount means the 2004-11 LBSF Settlement Amount plus the 2006-05 LBSF Settlement Amount.

Mahogany means Mahogany Capital Limited ACN 107 882 753 (in Liquidation).

Mahogany Noteholders means the holders from time to time of notes issued by Mahogany and known as the Mahogany Notes Series I and the Mahogany Notes Series II.

Permitted Disclosures means the disclosures in or substantially in the terms set out in Schedule 1 to this deed.

Perpetual Settlement Amount means the 2004-11 Perpetual Settlement Amount plus the 2006-05 Perpetual Settlement Amount.

Principal Parties means Perpetual and LBSF.

Settlement Payment Deed

Proceedings means any proceedings, suit or action arising out of or in connection with any Disputes or otherwise arising out of or in connection with this deed (including regarding its existence, validity or termination).

Related Body Corporate has the meaning given to that term in section 50 of the *Corporations Act 2001* (Cth).

Saphir means Saphir Finance Public Limited Company, a public limited company incorporated in Ireland whose registered office is at AIB International Centre, International Financial Services Centre, Dublin 1, Ireland.

Termination and Settlement Agreement means the document so entitled dated on or about the date of this deed between, among others, Perpetual, LBSF and BNY, substantially in the form annexed hereto (**Schedule 2**).

2004-11 LBSF Settlement Amount means

2004-11 Perpetual Settlement Amount means the 2004-11 Termination and Redemption Payments minus the 2004-11 LBSF Settlement Amount.

2006-05 LBSF Settlement Amount means

2006-05 Perpetual Settlement Amount means the 2006-05 Termination and Redemption Payments minus the 2006-05 LBSF Settlement Amount.

1.2 Interpretation

The following apply in the interpretation of this deed, unless the context requires otherwise.

- (a) A reference to this deed, this document or a similar term means either the agreement set out in this document or the document itself, as the context requires.
- (b) A reference to any statute, regulation, rule or similar instrument includes any consolidations, amendments or re-enactments of it, any replacements of it, and any regulation or other statutory instrument issued under it.
- (c) A reference to the singular includes the plural number and vice versa.
- (d) A reference to a gender includes a reference to each gender.
- (e) A reference to a party means a person who is named as a party to this deed.
- (f) **Person** includes a firm, corporation, body corporate, unincorporated association and a governmental authority.
- (g) A reference to a party or a person includes that party's or person's executors, legal personal representatives, successors, liquidators, administrators, trustees in bankruptcy and similar officers and, where permitted under this deed, their substitutes and assigns.

Settlement Payment Deed

- (h) **Includes** means includes but without limitation.
- (i) Where a word or expression has a defined meaning, its other grammatical forms have a corresponding meaning.
- (j) A reference to doing something includes an omission, statement or undertaking (whether or not in writing) and includes executing a document.
- (k) A reference to a clause, schedule or annexure is a reference to a clause of, or a schedule or an annexure to this deed.
- (I) A reference to time is to the time in the place where a thing is to be done, unless specified otherwise.
- (m) A reference to dollars or \$ is to Australian currency.
- (n) A heading is for reference only. It does not affect the meaning or interpretation of this deed.
- (o) Any schedule attached to this deed forms part of it. If there is any inconsistency between any clause of this deed and any provision in any schedule, the clause of this deed prevails.

1.3 Limitation of liability

- (a) Perpetual enters into this deed in the following capacities and in no other capacity:
 - note trustee and security trustee in relation to the issue of notes by Mahogany and known as the Mahogany Notes Series I (Mahogany I Notes); and
 - (ii) note trustee and security trustee in relation to the issue of notes by Mahogany and known as the Mahogany Notes Series II (Mahogany II Notes).
- (b) A liability of Perpetual arising under or in connection with this deed in relation to the Mahogany I Notes or, as the case may be, the Mahogany II Notes, is limited to and can be enforced against it only to the extent to which it can be satisfied out of:
 - the amount held by Perpetual for the time being and from time to time in its capacity as trustee for the Mahogany I Notes or, as the case may be, the Mahogany II Notes out of which Perpetual is actually indemnified for the liability; and/or
 - (ii) the amount held by the Settlement Payment Agent from time to time to which Perpetual is entitled in accordance with the terms of this deed.

This limitation of liability applies despite any other provision of this deed and extends to all liabilities and obligations of Perpetual in any way connected with any representation, warranty, conduct, omission, indemnity, agreement or transaction made in or related to this deed.

2 Appointment of Settlement Payment Agent

2.1 Appointment

The Principal Parties appoint the Settlement Payment Agent with effect from the Effective Date as depository and custodian for the safekeeping of the Termination and Redemption Payments received by it pursuant to the Termination and Settlement Agreement and to hold such amounts received and any interest paid thereon in for payment to Perpetual and LBSF in accordance with the terms and conditions set out in this deed.

2.2 Acceptance of Appointment

The Settlement Payment Agent accepts its appointment on and subject to the terms and conditions set out in this deed.

3 Receipt and disbursement of the Termination and Redemption Payments

3.1 Receipt

- (a) Perpetual and LBSF have agreed under the Termination and Settlement Agreement for Saphir to pay or procure the payment of the Termination and Redemption Payments to the Settlement Payment Agent on the Actual Payment Date, subject to the terms of the Termination and Settlement Agreement.
- (b) The Settlement Payment Agent must give written notice to each Principal Party confirming receipt of the Termination and Redemption Payments within 1 Business Day of receipt.

3.2 Disbursement

Within 1 Business Day after receipt in cleared funds of the funds to be received under clause 3.1, the Settlement Payment Agent must:

- (a) first, pay to LBSF the LBSF Settlement Amount; and
- (b) second, pay each Perpetual Settlement Amount into a separate controlled moneys trust account (one corresponding to each Series) to be maintained by the Settlement Payment Agent and held for Perpetual.

3.3 **Disbursement – Confidentiality Windback Date**

Promptly following a written request by Perpetual sent by fax and email to the Settlement Payment Agent to do so (which request cannot be made with effect prior to the Confidentiality Windback Date), the Settlement Payment Agent must pay to Perpetual each Perpetual Settlement Amount, plus any interest accrued thereon. Perpetual shall simultaneously with the request to the Settlement Payment Agent send a copy of such request by fax and email to LBSF.

4 Legal costs of the UK Proceedings

4.1 Costs Settlement

LBSF shall pay to Perpetual the sum of £280,320.00 (**Costs Settlement**) in full and final settlement of the sums owing to Perpetual pursuant to paragraph 4 of the order of the Chancellor in the UK Proceedings dated 28 July 2009.

4.2 **Payment of Costs Settlement**

The Costs Settlement shall be paid to the following bank accounts, to be received in cleared funds within 3 Business Days of receipt by LBSF of the LBSF Settlement Amount.

As to £168,192.00 to:

Bank: Australia and New Zealand Banking Group Limited For: Perpetual Trustee Company Limited Account Name: Funds in the Series 1 Rectification Trust BSB: 012003 Account Number: 8370 60398

As to £112,128.00 to:

Bank: Australia and New Zealand Banking Group Limited For: Perpetual Trustee Company Limited BSB: 012003 Account Name: Funds in the Series 2 Rectification Trust BSB: 012003 Account Number: 8370 60419

5



Settlement Payment Deed HENRY DAVIS YORK

6 Duties of Settlement Payment Agent

6.1 **Duties and Responsibilities**

The duties and responsibilities of the Settlement Payment Agent are limited to those set out in this deed and, accordingly, the Settlement Payment Agent:

- (a) acts as a depository and custodian for the amounts it receives under this deed and the Termination and Settlement Agreement and any cash, distributions or interest received in connection therewith, only and is not responsible or liable for the sufficiency or validity of any funds deposited with it or the credit of the relevant account bank;
- (b) is not under any duty to inquire into the terms and provisions of any agreement or instruction other than as set out in this deed;
- (c) is not and must not be treated as having the duties or responsibilities of a trustee or fiduciary acting for the benefit of any Principal Party;
- (d) will have no further duties or responsibilities under this deed following payment of the payments referred to in clause 3;
- (e) is not required to use or advance its own funds or otherwise incur financial liability on its part in performance of its duties or the exercise of its rights under this deed; and
- (f) is not precluded from representing Perpetual or any Related Body Corporate as its solicitors in any matter, including in connection with a dispute under this deed, and LBSF shall not allege it is so precluded.

6.2 Notifications to Settlement Payment Agent

The Settlement Payment Agent:

- (a) is only subject to, and only authorised to recognise, notifications or directions given in accordance with this deed; and
- (b) is entitled to rely on any notification provided to the Settlement Payment Agent, which the Settlement Payment Agent in good faith believes to be genuine.

6.3 **Dispute between Principal Parties**

The Principal Parties must pursue any legal redress or recourse in connection with any dispute relating to the arrangements contained in this deed without making the Settlement Payment Agent a party, unless the dispute is based upon a breach of the Settlement Payment Agent's obligations under this deed, or the fraud, negligence or misconduct of the Settlement Payment Agent or any of its partners, employees, agents or representatives.

6.4 Release

- (a) The Settlement Payment Agent will not be liable for any error of judgement or for any acts done or steps taken or omitted by it in connection with this deed, except for the Settlement Payment Agent's breach of its obligations under this deed due to its own fraud, negligence or misconduct or that of any of its partners, employees, agents or representatives.
- (b) Each of the Principal Parties waives any claim which it may have or may assert against the Settlement Payment Agent arising out of the execution, delivery or performance by the Settlement Payment Agent of this deed, unless that claim is based upon a breach of the Settlement Payment Agent's obligations under this deed, or the fraud, negligence or misconduct of the Settlement Payment Agent or any of its partners, employees, agents or representatives.

6.5 Indemnity

Each of the Principal Parties severally indemnifies (and must keep indemnified) the Settlement Payment Agent, its partners, employees, agents and representatives (**Indemnified Persons**) against any and all liabilities (including professional and legal costs on a full indemnity basis) that the Settlement Payment Agent suffers or may suffer in relation to:

- (a) the performance of the Settlement Payment Agent's obligations under this deed;
- (b) the Settlement Payment Agent entering into or participating in this deed;
- (c) the Settlement Payment Agent commencing, defending, participating or being joined in any court proceedings, dispute or arbitration in relation to this deed or the performance of the parties obligations under this deed,

other than a liability due to a breach of the Settlement Payment Agent's obligations under this deed, or the fraud, negligence or misconduct of the Settlement Payment Agent or any of its partners, employees, agents or representatives.

6.6 **Fees**

In consideration of the Settlement Payment Agent performing its functions and duties under this deed, the Settlement Payment Agent shall be entitled to be paid a fee by Perpetual in the amount and at the times agreed from time to time with Perpetual.

6.7 No Set-Off

All payments under this deed must be made:

- (a) (manner of payment) in immediately available funds;
- (b) (time for payment) not later than 11.00 am (local time in the place of payment) on the due date;
- (c) (**direction for payment**) to the account specified or as the recipient otherwise directs from time to time for the purposes of this deed; and
- (d) (without counterclaim) without any counterclaim, deduction, withholding or set-off unless compelled by law.

7 Confidentiality

7.1 Confidentiality – LBSF and Perpetual

Each of LBSF and Perpetual hereby agrees with the other of them (**Other Party**) to use best endeavours to ensure that the existence of this deed and the terms and conditions of this deed (including, without limitation, the amounts payable hereunder) (the **Confidential Information**) are confidential and shall not be disclosed to any other person without the prior written consent of Other Party, in which circumstances disclosure shall, insofar as is practicable, be made only after consultation with the Other Party and after taking into account the reasonable requirements of the Other Party provided that nothing in this clause can be taken as preventing:

- (a) disclosure as expressly contemplated in, required by or permitted by this deed (if applicable);
- (b) disclosure of the existence of this deed in the Termination and Settlement Agreement;
- (c) disclosure of any material or information that is already in the public domain provided such material or information did not enter the public domain by way of a breach of this deed by the party making the disclosure;
- (d) disclosure as required by any applicable law, listing rule or authority of a court, administrative tribunal or arbitration panel, regulatory or supervisory body and/or their duly appointed representatives;
- (e) disclosure to the extent necessary to obtain any necessary approval from a court with jurisdiction over either of the Principal Parties to enter into this deed or to enter into the Termination and Settlement Agreement;

Settlement Payment Deed

- disclosure to a party's accountants, auditors, legal or other professional advisers and/or insurers which, in each case, are themselves subject to a duty of confidentiality;
- (g) disclosure to the extent necessary to perform or enforce this deed and/or the Termination and Settlement Agreement in accordance with their terms;
- (h) disclosure to each party's respective directors, officers, employees and (in the case of LBSF) Affiliates or (in the case of Perpetual) Related Bodies Corporate, in each case, solely for the purpose of effecting this deed (it being understood that such parties will be informed of the confidential nature of the deed and its terms and conditions);
- disclosure (in the case of LBSF only) to the court-sanctioned committee of unsecured creditors representing the creditors of LBSF (and other Affiliates) and the committee's respective advisers;

(k) the Permitted Disclosures.

For these purposes, LBSF agrees that the initial implementation of the Confidentiality Protocol satisfies Perpetual's best endeavours obligations under this clause 7.

7.2 Confidentiality – Settlement Payment Agent

The Settlement Payment Agent hereby agrees to use best endeavours to ensure that the existence of this deed and the terms and conditions of this deed (including, without limitation, the amounts payable hereunder) are confidential and shall not be disclosed to any other person without the prior written consent of Perpetual and LBSF, in which circumstances disclosure shall, insofar as is practicable, be made only after consultation with the other parties and after taking into account the reasonable requirements of Perpetual and LBSF provided that nothing in this clause can be taken as preventing:

- (a) disclosure as expressly contemplated in, required by or permitted by this deed (if applicable);
- (b) disclosure of any material or information that is already in the public domain provided such material or information did not enter the public domain by way of a breach of this deed by the Settlement Payment Agent;
- disclosure as required by any applicable law, listing rule or authority of a court, administrative tribunal or arbitration panel, regulatory or supervisory body and/or their duly appointed representatives;

- (d) disclosure to the Settlement Payment Agent's partners, directors, officers, employees and Related Bodies Corporate, in each case, solely for the purpose of effecting this deed (it being understood that such parties will be informed of the confidential nature of the deed and its terms and conditions);
- (e) disclosure to the Settlement Payment Agent's accountants, auditors, legal or other professional advisers and/or insurers which, in each case, are themselves subject to a duty of confidentiality; and
- (f) disclosure to the extent necessary to perform or enforce this deed and/or the Termination and Settlement Agreement in accordance with their terms.

7.3 Ensure compliance

- (a) Each party shall use its best endeavours to ensure that any partner, employee, director, officer, representative or adviser to whom it discloses another party's Confidential Information will comply with the provisions of this clause 7.
- (b) If either Perpetual or the Settlement Payment Agent become aware of disclosure of Confidential Information in breach of the provisions of this clause 7, they must:
 - (i) notify the other parties of such disclosure;
 - take such action as is necessary and within its respective power to take to enforce their rights against the individual who disclosed Confidential Information in breach of the provisions of this clause 7, including taking legal action to seek to prevent further disclosure; and
 - (iii) take all reasonable actions to retract, or seek the retraction of, such disclosure.

For the purposes of this clause 7.3(b), Perpetual and the Settlement Payment Agent shall only be deemed to be aware of a disclosure to which this clause applies if one of the addressees of the Confidentiality Protocol becomes so aware.

(c) Perpetual acknowledges that damages may not be an adequate remedy for a breach of the terms of this clause 7 and LBSF is entitled to seek the remedies of injunction, specific performance and other equitable relief for a threatened or actual breach and that nothing contained herein will be construed as prohibiting LBSF from pursuing any other remedies available to it, either at law or in equity, for such breach or threatened breach.

Settlement Payment Deed

8 Counterparts

If this deed is signed in counterparts then each counterpart is an original and together they constitute one deed.

9 Entire agreement

This deed and, in the case of the Principal Parties only, the Termination and Settlement Agreement:

- (a) records the entire agreement between the parties; and
- (b) supersedes all previous negotiations, understandings, representations and agreements,

in relation to the subject matter of this deed.

10 Further assurance

Each party must promptly do anything (such as obtaining consents, signing and producing documents, getting documents completed and signed and filing documents with a court) which is reasonably required by another party and which that other party reasonably considers necessary for the purposes of giving effect to or completing the provisions of this deed or any transaction contemplated by it.

11 Representations and Warranties

Each party hereto makes the following representations and warranties in relation to it on the Effective Date to each other party:

- (a) it is duly incorporated, established or constituted (as the case may be) and validly existing under the laws of its country of incorporation, establishment or constitution (as the case may be);
- (b) it has the power to enter into and perform, and has taken all necessary corporate action to authorise the entry into, performance and delivery of this deed and the transactions contemplated by this deed;
- (c) this deed constitutes its legally binding, valid and enforceable obligation;
- (d) the entry into and performance by it of, and the transactions contemplated by, this deed do not and will not conflict in any material respect with any law or regulation or judicial or official order, its constitutional documents or any document which is binding upon it or its assets; and

(e) all authorisations required in connection with the entry into, performance, validity and enforceability of, and the transactions contemplated by, this deed have been obtained or effected (as appropriate) and are in full force and effect.

12 Governing law

12.1 Governing law

This deed is governed by and is to be construed in accordance with the laws of New South Wales.

12.2 Jurisdiction

Each party irrevocably agrees that:

- (a) the courts of New South Wales have exclusive jurisdiction to hear and determine any Proceedings and to settle any Disputes and each party irrevocably submits to the exclusive jurisdiction of the courts of New South Wales; and
- (b) any judgment in Proceedings taken in the courts of New South Wales shall be conclusive and binding on it and may be enforced in any other jurisdiction.

12.3 Forum non conveniens

Each party also irrevocably waives (and irrevocably agrees not to raise) any objection which it might at any time have on the ground of *forum non conveniens* or on any other ground to Proceedings being taken in the courts of New South Wales

12.4 Benefit

This jurisdiction agreement is not concluded for the benefit of only one party.

Settlement Payment Deed

Legal costs

13.1 **Costs**

13

The parties must pay their own legal costs and disbursements in connection with this deed.

14 Notices

14.1 Requirements

Any notice or other communication required to be given by this deed before a right can be exercised (**notice**) must be:

- (a) in legible writing;
- (b) subject to clause 14.5, signed by the party giving it (sender) or by its authorised representative or, in the case of LBSF, by LAMCO Services Ltd or LBHI Estates Ltd; and
- (c) delivered by hand or sent by post (air mail if sent to an address in another country), sent to the relevant fax number or emailed to the relevant email address set out in the Key Details.

14.2 Change of address

A party may change its address, fax number or email address for the purpose of notices by giving notice of that change to each other party in accordance with the provisions of clause 14.1.

14.3 When given

Notices are taken to be given:

- (a) in the case of delivery by hand, when delivered;
- (b) in the case of delivery by post, on the third (seventh, if sent to an address in another country) Business Day after the date of posting;
- (c) in the case of delivery by fax, at the time shown on a transmission report by the machine from which the fax was sent which indicates that the fax communication was sent at that time, in its entirety and without error to the fax number of the recipient; and
- (d) in the case of delivery by email, when sent, unless the sender is notified, by a system or person involved in the delivery of the email, that the email was not successfully sent.

14.4 Fax or email

If a notice by fax or email is given:

- (a) on a day which is not a Business Day in the place in which the fax or email is received, or
- (b) after 4.00 pm (local time) on a day which is a Business Day in the place in which the fax or email is received,

the notice will be taken to have been given at the commencement of business on the next Business Day in the place in which the fax or email is received.

14.5 Email signing

For the purposes of clause 14.1, a notice by email is taken to be signed by the sender if the notice is in the form of a signed document in Portable Document Format (pdf) and attached to the email.

14.6 **Process agent**

- LBSF irrevocably appoints Jones Day, Sydney office, to be its process agent to accept on its behalf service of any writ, summons, claim form, order, judgment or any other document relating to or in connection with any Proceedings in New South Wales (each a Service Document).
- (b) LBSF irrevocably agrees that any Service Document may be sufficiently and effectively served on it in connection with Proceedings in New South Wales by service on the agent specified above.

15 Waiver

15.1 No waiver

A party does not waive a right or remedy in connection with this deed if it:

- (a) fails to exercise its right or remedy;
- (b) only partially exercises the right or remedy; or
- (c) delays in exercising the right or remedy.

15.2 Maintenance of rights

A party which exercises a single right or remedy or partially exercises a right or remedy maintains its right to:

- (a) further exercise the right or remedy; or
- (b) exercise another right or remedy.

15.3 Waiver effective

A waiver is effective only if in writing and signed by or on behalf of the party to be bound and is effective to the extent that the party giving it expressly states in writing.

16 Severability

If any part of this deed (save for the obligation to make the payments set out in clause 3) is for any reason unenforceable, that part must be read down to the extent necessary to preserve its operation. If it cannot be read down, it must be severed.

17 Payment

17.1 **To Perpetual**

All sums payable to Perpetual pursuant to this deed shall be made to the following bank accounts:

2004-11 Perpetual Settlement Amount (and any interest thereon)

Bank: Australia and New Zealand Banking Group Limited For: Perpetual Trustee Company Limited Account Name: Funds in the Series 1 Rectification Trust BSB: 012003 Account Number: 8370 60398

2006-5 Perpetual Settlement Amount (and any interest thereon)

Bank: Australia and New Zealand Banking Group Limited For: Perpetual Trustee Company Limited Account Name: Funds in the Series 2 Rectification Trust BSB: 012003 Account Number: 8370 60419

17.2 To LBSF

All sums payable to LBSF pursuant to this deed shall be made to the following bank account:

National Australia Bank Limited, Melbourne			
SWIFT:	NATAAU33033		
For:	US Bank, Minneapolis, MN		
SWIFT:	USBKUS44FEX		
a/c:	1803005789500		
Attn:	STL FX		
For further credit to:	Lehman Brothers Special Financing Inc.		

Settlement Payment Deed HENRY DAVIS YORK

Reference: Perpetual/Saphir 2004-11/2006-05

18 Time

If any provision of this deed requires a party to perform an obligation under this deed and a time is prescribed for the performance of the obligation, then time is of the essence in respect of the performance of the obligation.

Settlement Payment Deed	HENRY DAVIS YORK	
EXECUTION		
Executed as a deed on	2010	
Signed, sealed and delivered by Perpetual Trustee Company Limited by its attorney under Power of Attorney dated		
who is personally known to me:		
Signature of witness	Signature of attorney	
Print name	Print name By executing this document the attorney states that the attorney has received no notice of revocation of the Power of Attorney	
Address		
EXECUTED AND DELIVERED AS A DEED BY LEHMAN BROTHERS SPECIAL FINANCING)	
By:)	
Signature:)	
Signed, sealed and delivered for and on behalf of Henry Davis York:		
Signature of witness	Signature of partner	
Print name	Print name	

Proposed Order

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF NEW YORK			
In re	:	Chapter 11 Case No.	
LEHMAN BROTHERS HOLDINGS INC., et al.,	:	08-13555 (JMP)	
Debtors.	:	(Jointly Administered)	
	X		

ORDER PURSUANT TO RULE 9019 OF THE FEDERAL RULES OF BANKRUPTCY PROCEDURE FOR APPROVAL OF A SETTLEMENT AMONG LEHMAN BROTHERS SPECIAL FINANCING INC., BNY CORPORATE TRUSTEE SERVICES LIMITED, PERPETUAL TRUSTEE COMPANY LIMITED, AND OTHERS, RELATING TO CERTAIN SWAP <u>TRANSACTIONS WITH SAPHIR FINANCE PUBLIC LIMITED COMPANY</u>

Upon the motion, dated November 24, 2010 (the "<u>Motion</u>"), of Lehman Brothers Special Financing Inc. ("<u>LBSF</u>") and its affiliated debtors in the above-referenced chapter 11 cases, including Lehman Brothers Holdings Inc. ("<u>LBHI</u>"), as debtors and debtors-in-possession (collectively, the "<u>Debtors</u>" and, together with their non-debtor affiliates, "<u>Lehman</u>"), pursuant to Rule 9019 of the Federal Rules of Bankruptcy Procedure (the "<u>Bankruptcy Rules</u>") for approval of a Settlement⁵ among LBSF, LBHI, BNY Corporate Trustee Services Limited, Perpetual Trustee Company Limited, and others, relating to certain swap transactions with Saphir Finance Public Limited Company, all as more fully described in the Motion; and the Court having jurisdiction to consider the Motion and the relief requested therein in accordance with 28 U.S.C. §§ 157 and 1334 and the Standing Order M-61 Referring to Bankruptcy Judges for the Southern District of New York Any and All Proceedings Under Title 11, dated July 10, 1984 (Ward, Acting C.J.); and consideration of the Motion and the relief requested therein being a core

5

Capitalized terms not otherwise defined herein shall have the meaning ascribed to them in the Motion.

proceeding pursuant to 28 U.S.C. § 157(b); and venue being proper before this Court pursuant to 28 U.S.C. §§ 1408 and 1409; and due and proper notice of the Motion having been provided to (i) the U.S. Trustee; (ii) the attorneys for the Creditors' Committee; (iii) the Securities and Exchange Commission; (iv) the Internal Revenue Service; (v) the United States Attorney for the Southern District of New York; and (vi) all other parties entitled to notice in accordance with the procedures set forth in the second amended order entered on June 17, 2010 governing case management and administrative procedures for these cases [Docket No. 9635]; and a hearing (the "<u>Hearing</u>") having been held to consider the relief requested in the Motion; and the Court having found and determined that the relief sought in the Motion is in the best interests of the Debtors, their estates and creditors, and all parties in interest and that the legal and factual bases set forth in the Motion establish just cause for the relief granted herein; and after due deliberation and sufficient cause appearing therefor, it is

ORDERED that the Motion is granted; and it is further

ORDERED that, pursuant to Bankruptcy Rule 9019, the Debtors' execution and performance of the Termination and Settlement Agreement and the Settlement Payment Deed is approved; and it is further

ORDERED that LBSF and LBHI are authorized to execute, deliver, implement and fully perform any and all obligations, instruments, documents and papers and to take any and all actions reasonably necessary or appropriate to consummate the Termination and Settlement Agreement and the Settlement Payment Deed and perform any and all obligations contemplated therein; and it is further

ORDERED that the terms of this Order shall be immediately effective and enforceable upon its entry; it is further

2

ORDERED that notice of the Motion as provided therein shall be deemed good and sufficient notice of such Motion; and it is further

ORDERED that this Court retains jurisdiction with respect to all matters arising

from or related to the implementation of this Order.

Dated: _____, 2010 New York, New York

UNITED STATES BANKRUPTCY JUDGE